Changes in body composition over 5 years after Roux-en-Y Gastric Bypass and Sleeve Gastrectomy.

Bradley Matheson

A thesis submitted in partial fulfilment of the requirements for the degree of Master of Health Sciences in Nutrition and Dietetics, The University of Auckland, 2024.

Abstract

Background and aim: Bariatric surgery produces extensive weight loss unrivalled by non-surgical obesity treatments. However, there is limited evidence comparing the effects of different procedures on body composition beyond the first year. Findings are often further limited by body composition assessment methods with unreliable or unproven validity in individuals with obesity.

This thesis aimed to compare changes in body composition over five years following two types of bariatric surgery: Roux-en-Y gastric bypass (RYGB) and sleeve gastrectomy (SG). Secondly, it aimed to assess the validity of bioimpedance techniques for measuring body composition in a population with obesity using dual-energy x-ray absorptiometry (DXA) as a reference standard.

Methods: This thesis analysed data collected from participants randomised to undergo either RYGB or SG. Data was collected two days before surgery, one year after surgery, and five years after surgery. Body composition was measured using DXA, single-frequency bioelectrical impedance analysis (SF-BIA) and multi-frequency bioimpedance analysis (MF-BIA). Assessment outcomes included body weight, fat-free mass, appendicular skeletal muscle mass, fat mass, visceral adipose tissue mass, and android adipose tissue mass.

Results: Body composition assessment information was available for 91 participants. Compared to the SG group at five years, the RYGB group had more significant reductions in body weight measured as percentage excess weight loss (%EWL) (p<0.0001), total body fat (p<0.0001), body fat percentage (p=0.002), android fat mass (p<0.001), and visceral fat mass (p<0.001), and a more significant increase in fat-free mass percentage (p=0.003). Fat mass and fat-free mass derived from five BIA equations were compared against DXA measurements. Bias ranged from -8.2% to 10.23% of DXA values, and Pearson correlation coefficients exceeded 0.94. However, the limits of agreement were large for all five equations.

Conclusions: Greater weight loss was achieved following RYGB due to greater initial weight loss and lower weight regain. Qualitative changes in body composition appear similar between the groups, and differences are likely associated with the extent of weight loss.

Factory BIA equations and equations extracted from pre-existing literature failed to estimate body composition in a population with obesity accurately. Without derivation and validation of specific equations for obesity, SF-BIA & MF-BIA applications remain limited by poor individual accuracy and large limits of agreement.

Acknowledgements

First and foremost, I express my sincere appreciation and gratitude to my supervisor, Associate Professor Lindsay Plank. Thank you for your invaluable support, plentiful wisdom and endless patience. I am truly honoured to have had the opportunity to experience academia under your tutelage.

Thank you to all those coordinating and delivering the Nutrition and Dietetics course. You provided a steppingstone in my journey that I will never forget. And to my classmates, thank you for the laughs, smiles and even collective sighs over the last two years. I am excited to cross paths again during our dietetic careers.

I would also like to thank Associate Professor Alexander Tups from the University of Otago. My experience while previously studying under your supervision was invaluable to my personal and academic growth. Your direct and sincere approach enabled me to think bigger and aim higher.

To my friends from home and abroad, thank you. You were all critical in enabling me to complete this thesis and to get here in the first place. Laurence, I am forever grateful for the few months I was welcomed into your house, which truly felt like home. Carl, thank you for always being there for my endless rants, regardless of the distance. And to my amazing partner Samantha, thank you for sticking by my side through high stresses and long distances. Your perfectionism motivates me to push above and beyond.

And to my parents, Roslyn and Neil, calling you supportive parents would be a disservice. There are no words to convey my gratitude and love for you. Thank you for giving me opportunities not afforded to you. My accomplishments are just as much yours as they are mine.

Lastly, a special thanks to my sister Lana-Marie. Thank you for encouraging me to pursue my passion and coincidentally follow in your footsteps. After years of bickering like siblings do, I am grateful that nutrition and dietetics (as well as maturing) has brought us closer together. I am eternally grateful to have you as a role model. Thank you for setting the standard so high, not only as a dietitian but as a person.

Table of Contents

Abstract	ii
Acknowledgements	iii
Table of Contents	iv
List of Tables	vii
List of Figures	viii
Chapter 1. Introduction	1
1.1 Introduction to Obesity	1
1.2 Thesis Objectives	2
1.3 Thesis Overview	
Chapter 2. Literature Review	4
2.1 Body Composition Assessment	4
2.1.1 Anthropometry	4
2.1.1.1 Body Mass Index	4
2.1.1.2 Waist Circumference	5
2.1.1.4 Skinfold Thickness	6
2.1.2 Bioimpedance	7
2.1.2.1 Bioelectrical Impedance Analysis - Single Frequency	7
2.1.2.2 Bioelectrical Impedance Analysis - Multi-frequency	8
2.1.2.3 Bioimpedance Spectroscopy	8
2.1.3 Direct Methods	9
2.1.3.1 Isotope Dilution	9
2.1.3.2 Total Body Counting	9
2.1.3.3 Neutron Activation	
2.1.4 Criterion Methods	
1.4.1 Hydrodensitometry	
1.4.2 Dual-Energy X-Ray Absorptiometry	11
1.1.3.3 Magnetic Resonance Imaging	
2.1.5 Comparison of DXA and Bioimpedance Techniques	
2.2 Bariatric Surgery and Body Composition	
2.2.1 Bariatric Surgery	
2.2.2 Sleeve Gastrectomy	
2.1.2 Roux-en-Y Gastric Bypass	
2.2.3 Weight Loss	17
2.2.4 Fat Mass	17
2.2.5 Fat Mass Distribution	
2.2.6 Fat-Free Mass	19

2.2.7 Tuno 2 Dishotos Mollitus Domission	24
Charter 2. Matheda	
Chapter 3. Methods	
3.1 Randomised Clinical Trial	
3.1.1 Etnics and Trial Registration	
3.1.2 Power Analysis	
3.1.3 Patient eligibility and recruitment.	
3.1.4 Study Design	
3.2 Study Protocol	
3.2.1 Very Low-Calorie Diet (VLCD)	
3.2.2 Randomisation and Blinding	
3.2.3 Surgical Procedures	27
3.2.3.1 Laparoscopic Sleeve Gastrectomy	28
3.2.3.2 Laparoscopic Silastic Ring Roux-en-Y Gastric Bypass	28
3.2.4 Post-Operative Management	28
3.3 Outcome Measures	28
3.3.1 Anthropometry	28
3.3.1.1 Body Weight	
3.3.1.2 Height	
3.3.1.3 Anthropometric Derivatives	29
3.3.2 Dual-energy X-ray Absorptiometry	29
3.3.3 Bioimpedance Assessment	29
3.3.3.1 Single-Frequency Bioimpedance Analysis	29
3.3.3.2 Multi-frequency Bioimpedance Analysis	
3.4 Statistical Analysis	30
Chapter 4. Results	
4.1 Patient Characteristics	31
4.2 Body Weight	33
4.3 Fat Mass	35
4.4 Fat-Free Mass	37
4.5 Bioimpedance Equation Performance	40
Chapter 5. Discussion	
5.1 Body Weight	48
5.2 Fat-free mass	50
5.3 Fat Mass	51
5.4 Fat Mass Distribution	51
5.5 Bioimpedance Equations	52
Chapter 6. Conclusion	53

	4
5	5

List of Tables

Table 1. Classification of adults according to BMI. 5
Table 2. Sex-specific waist circumference and risk of metabolic complications associated with
obesity in Caucasians
Table 3. Summary of Studies comparing body composition measured by bioimpedance techniques
to reference methods
Table 4. Summary of studies comparing body composition outcomes 1-year after SG and RYGB.21
Table 5. Summary of studies comparing body composition outcomes ≥3-years after SG and RYGB
Table 6. Baseline demographic characteristics
Table 7. Body weight and body composition measured by DXA at baseline, 1 year and 5 years32
Table 8. Change in BW and %EWL over 5 years 34
Table 9. Changes in total body fat and central fat mass over 5 years
Table 10. Changes in fat-free mass and appendicular skeletal muscle mass over 5 years
Table 11. Bioimpedance equations for body composition
Table 12. Fat mass and fat-free mass obtained by DXA and BIA equations
Table 13. Comparison of fat mass and fat-free mass obtained by DXA and BIA equations

List of Figures

Figure 1. Changes in body composition at 5 years following Roux-en-Y gastric bypass and sleeve
gastrectomy
Figure 2. Changes in body weight over 5 years following Roux-en-Y gastric bypass and sleeve
gastrectomy
Figure 3. Changes in total body fat over 5 years following Roux-en-Y gastric bypass and sleeve
gastrectomy
Figure 4. Changes in visceral adipose tissue and android adipose tissue over 5 years following
Roux-en-Y gastric bypass and sleeve gastrectomy
Figure 5. Changes in fat-free mass and appendicular skeletal muscle mass over 5 years following
Roux-en-Y gastric bypass and sleeve gastrectomy
Figure 6. Bland-Altman plots and scatter graphs comparing fat mass and fat-free mass measured by
DXA and bioimpedance analysis derived values for five equations47

Chapter 1. Introduction

1.1 Introduction to Obesity

Obesity is a chronic disease defined by excessive body fat that poses a risk to health (World Health Organisation, 2024). The distinction between the categorisation of overweight and obesity is made using body mass index (BMI) cutoffs of 25 kg/m² and 30 kg/m² (World Health Organisation, 2024). Despite mass recognition of the obesity epidemic, the prevalence of obesity continues to increase globally, with an estimated third of the world's population classified as overweight or obese (Chooi et al., 2019; GBD 2015 Obesity Collaborators et al., 2017). Obesity is a multisystem disease and a prominent risk factor for type 2 diabetes mellitus (T2DM), cardiovascular disease, and various types of cancer (Guh et al., 2009). Obesity is emerging as a major economic burden as increased BMI is correlated with reductions in productivity, early mortality, and higher per capita healthcare costs (Dee et al., 2014).

Strategies for treating obesity are as fundamentally simplistic as the origins of obesity itself. Fat deposition occurs during states of positive energy balance where energy intake exceeds energy expenditure (Hill, 2006). Therefore, treatment strategies address one or both components of the equation by increasing expenditure or decreasing caloric intake. Current non-surgical treatment strategies include dietary modification, physical activity, multi-faceted lifestyle modifications and pharmacotherapy (Blüher et al., 2023). However, non-surgical interventions consistently produce only modest reductions in body weight (Ruban et al., 2019). Behavioural modification and public health interventions contribute greatly to obesity prevention, but the efficacy of bariatric surgery for treating excessive adiposity is currently unrivalled. Increased acceptance of surgical interventions for treating obesity is reflected in the steadily increasing volume of bariatric procedures performed (Chousleb et al., 2019; Clapp et al., 2022).

1.2 Thesis Objectives

The number of bariatric surgeries performed in New Zealand has grown significantly in both public and private settings as a treatment for obesity. The sleeve gastrectomy (SG) and gastric bypass are the most common procedures and result in drastic changes in body composition. Both locally and internationally, the gastric sleeve has replaced the gastric bypass as the primary procedure. However, evidence supporting either procedure is limited. Previous comparative investigations are from unrandomised cohorts and infrequently extend beyond one year post-operatively. Therefore, the first objective of this thesis was to;

1) Compare body composition changes, including fat mass (FM), visceral adipose tissue (VAT), subcutaneous adipose tissue, fat-free mass (FFM) and appendicular skeletal muscle mass (ASMM) over 5 years following laparoscopic banded-Roux-en-Y gastric bypass (RYGB) and laparoscopic SG.

Despite the value of information obtained from body composition analysis to researchers and health professionals, it is infrequently utilised. Methodologies with sufficient evidence supporting their accuracy, such as dual-energy x-ray absorptiometry (DXA) and four-compartment models, are impractical for widespread use due to their low availability and higher costs. Bioimpedance devices offer an affordable and accessible method of assessing body composition. However, the validity of bioimpedance techniques is often contested. Therefore, the next objective of this thesis was to:

2) Compare the validity of two bioimpedance methods for assessing FM and FFM using DXA as the reference technique.

The accuracy of single-frequency bioelectrical impedance analysis (SF-BIA) and multifrequency bioelectrical impedance analysis (MF-BIA) is heavily dependent on the validity of the prediction equation used. Findings from frequently used equations, such as from large studies or those provided by manufacturers, should be cautiously interpreted. Prediction equations require validation in new populations due to inter-individual differences. The lack of population-specific equations is apparent in populations classified as overweight or obese. Bioimpedance techniques typically overestimate FFM and underestimate FM in these populations due to increased FFM hydration compared to normal-weight individuals. Generating an equation or validating existing equations for a population with obesity may improve the validity of SF-BIA in these cohorts. Therefore, the final objective of this thesis was to:

3) Extract from the published literature SF-BIA equations featured in validation studies of populations classified as overweight/obese and those derived from populations classified as

overweight/obese and compare body composition estimates generated by those equations with those from DXA in the current cohort of New Zealand bariatric patients.

1.3 Thesis Overview

Chapter 1 provides background information on the obesity epidemic and currently available treatment strategies.

Chapter 2 of this thesis is divided into two sections that review the current literature relating to body composition analysis methods and bariatric surgery outcomes. The first section covers different methodologies for assessing body composition and considers their application in a population with obesity. Section two reviews body composition outcomes following RYGB and SG. This section compares post-surgical changes in body weight, FM, FM distribution, FFM, skeletal muscle mass and T2DM management.

Chapter 3 describes the study methodology, including participant eligibility and recruitment, the study protocol, and descriptions of assessed outcomes. The techniques used for measuring body composition are also presented here.

Chapter 4 reports the results of this study.

Chapter 5 discusses the study results and compares them to pre-existing literature. The strengths and limitations of the study are discussed within the context of the study findings. Finally, recommendations for the application of the study findings and the direction of future research are discussed.

Chapter 2. Literature Review

2.1 Body Composition Assessment

Body composition refers to the components of body mass and their proportions (Müller et al., 2016). Body composition can be described using multi-compartment models ranging from elemental constitution at an atomic level to proportions of skeletal muscle and adipose tissue at the organ-tissue level (Müller et al., 2016). These models are the theoretical foundation for modern body composition analysis techniques. Body composition assessment techniques are utilised in clinical settings for the risk assessment of nutritional such as sarcopenia, obesity, and osteoporosis (Andreoli et al., 2016).

Body composition analysis methods can be classified as direct, indirect or criterion methods. Direct measures like isotope dilution and neutron activation and criterion methods like densitometry and DXA have represented 'gold standards' as they measure the body or its properties directly. However, these techniques are limited in clinical use as they require highly skilled operators and expensive equipment with low availability. In contrast, indirect methods see widespread clinical use as measurements are easy to obtain and require equipment that is affordable and readily available.

Indirect methods describe the relationship between proxy measures and body composition. Indirect methods such as BIA and anthropometry utilise regression equations derived from validation studies against direct or criterion methods to estimate body compartment components. Consequently, the precision of indirect methods is typically inferior and additional caution is required when applying equations beyond the original sample population.

2.1.1 Anthropometry

Anthropometric measurements are frequently used to describe body composition or body composition-related health risks. Anthropometric techniques are non-invasive, quick to obtain and use widely available equipment. Anthropometric measures commonly used to describe body composition include BMI and skinfold thickness (SFT) for total body fat (TBF), waist circumference for fat distribution, and mid-upper arm circumference (MUAC) for total muscle mass.

2.1.1.1 Body Mass Index

BMI is the most widely used anthropometric measure for assessing body fat. BMI is calculated by dividing a person's weight in kilograms by the square of their height in meters. BMI categorises

individuals by weight statuses that are associated with varying levels of disease risk (Table 1) (World Health Organisation, 2000). The primary limitation of BMI is that it is unable to distinguish between FM and FFM or quantify their distribution. The distribution of FM is of particular significance as central obesity is more strongly associated with mortality than BMI (Sahakyan et al., 2015).

Classification	BMI (kg/m ²)	Risk of comorbidities
Underweight	<18.5	Low*
Normal Range	18.5-24.99	Average
Overweight:	≥25	
Pre-obese	25-29.99	Increased
Obese Class I	30-34.99	Moderate
Obese Class II	35-39.99	Severe
Obese Class III	≥40	Very Severe

Table 1. Classification of adults according to BMI.

*Risk of other clinical problems is increased. BMI = Body mass index.

2.1.1.2 Waist Circumference

Waist circumference is an anthropometric measurement used as an indicator for intra-abdominal FM (Chan et al., 2003) and metabolic health risk (Darsini et al., 2020). Sex-specific thresholds of WC are used to categorise the risk of metabolic complications as either 'increased' or 'substantially increased' (Table 2)(World Health Organization, 2011). WC is strongly correlated with several health risks, such as type 2 diabetes mellitus, hypercholesterolemia, and hypertension (Darsini et al., 2020). Being able to describe body fat distribution patterns is the main advantage of using WC compared to BMI. However, waist circumference is better suited to describe disease risk than body composition compared to the array of quantitative body composition analysis (BCA) methods currently available.

Table 2. Sex-specific waist circumference and risk of metabolic complications associated with obesity in Caucasians.

Risk of metabolic	Waist Circumference (cm)				
complications	Men	Women			
Increased	≥ 94	≥ 80			
Substantially Increased	≥ 102	≥ 88			

MUAC is an anthropometric measurement used in malnutrition screening as an alternative to BMI (Benítez Brito et al., 2016; Van Tonder et al., 2019). In addition to its practicality for bedside assessment, it is generally accepted that MUAC is influenced to a lesser degree by hydration status and conditions of fluid overload than weight-based measures (Modi et al., 2015; Todorovic et al., 2011) Therefore, MUAC may be more reliable than BMI for monitoring changes in nutritional status in disease states with altered hydration states, such as heart failure, kidney disease, and obesity (Cotter et al., 2008; Gonçalves et al., 2006; Waki et al., 1991). MUAC can also be used to calculate mid-arm muscle area (MAMA) and then estimate total body muscle mass (Heynsfield et al., 1982). This is of particular interest for use in immobile or critically ill patients as total muscle mass can be estimated from measures done at bedside (MUAC & triceps SFT).

2.1.1.4 Skinfold Thickness

SFT is an anthropometric measurement obtained by pinching the skin with specialised callipers (Durnin & Rahaman, 1967). SFT measures subcutaneous body fat and enables the estimation of TBF through previously validated prediction equations. The age and sex-specific equations developed by Durnin and Womersley may be the most widely used (Durnin & Womersley, 1974). The sum of SFT measurements at up to four sites is related to body density measures in healthy subjects obtained by hydrodensitometry. FM can then be calculated from body weight using the Siri equation (Siri, 1956). Like other anthropometric techniques, the accuracy and repeatability of skinfold measurements are dependent on operator experience and the use of standardised methodology (Lohman et al., 1988; Totosy de Zepetnek et al., 2021).

Despite ongoing refinement of equations for estimating FM from SFT, their accuracy proves highly inconsistent when compared to modern 'gold standard' techniques like DXA (Bacchi et al., 2017; Benito et al., 2019; López-Taylor et al., 2018; Truesdale et al., 2016). The inconsistent performance of SFT equations is likely attributable to differences between study populations and the population from which the equation was derived. FM estimations from SFT rely on the assumption that body fat distribution is uniform. However, body fat distribution is shown to differ across age, ethnicity, sex, and body type (Hattori et al., 1991). Additionally, standard callipers may be unable to accommodate larger skinfold thicknesses seen in subjects with obesity (Gray et al., 1990).

2.1.2 Bioimpedance

Bioimpedance techniques involve measuring the resistance of the body to a weak, alternating electrical current (Kyle et al., 2004a). Whole-body resistance is typically measured using a current applied between electrodes on the ankle and wrist. Resistance is influenced by body composition as tissues with a high water content, like muscle, conduct current well, and tissues with low water content, like fat and bone, are poor conductors. When expressed as height (H)²/R, a derivative from the equation for the volume of a cylindrical conductor of electrolyte with resistance, R, a strong correlation is found with fluid volume measures from dilution experiments (Kushner, R. et al., 1992). However, modelling the body as a single cylinder or series of cylinders is a crude approximation. Therefore, regression equations based on the relationship between H²/R and fluid volume include adjustments for factors such as weight, sex, and age (Kyle et al., 2004a). Different applications of the basic principles behind bioimpedance and the utilisation of various other assumptions have led to the distinction of three main bioimpedance techniques. These techniques are SF-BIA, MF-BIA, and Bioimpedance Spectroscopy (BIS).

2.1.2.1 Bioelectrical Impedance Analysis - Single Frequency

Early bioimpedance devices measured bodily resistance using a single frequency. This application is appropriately named SF-BIA and typically uses a 50 kHz frequency. Resistance measures from SF-BIA are used to estimate total body water (TBW) through regression equations developed from reference data such as isotope dilution experiments. FFM and FM estimates can be derived from TBW and body weight, as FFM is assumed to have constant hydration at ~73% (Pace & Rathbun, 1945; Wang, Z. et al., 2000). Alternatively, some regression equations may estimate FM or FFM directly from total body resistance. Equations derived for body composition assessment by SF-BIA are presented in reviews by Kyle et al. (2004) and Houtkooper et al. (1996).

SF-BIA is an attractive technique as devices are easily accessible, affordable, and relatively simple to operate. However, the assumptions present in the methodology of SF-BIA introduce significant barriers to clinical application. Firstly, the assumption that resistance values at a 50 kHz frequency are representative of both ICW and ECW may be invalid outside of healthy populations (De Lorenzo et al., 1997). Unlike at high frequencies, current at intermediate frequencies only propagates through a portion of the intracellular fluid (Matthie, J. et al., 1998). In healthy populations, the proportions of extra and intracellular water are tightly controlled, and both ECW and TBW can be estimated with relative precision (Kushner, Robert, 1992). However, notable inaccuracies are observed in populations prone to altered hydration states (Kyle et al., 2004b). Additionally, SF-BIA estimates of TBW correlate poorly with reference measures in individuals

with severe obesity (Bernhard et al., 2016). This effect is likely attributable to altered tissue distribution and the correlation between the ECF:ICF ratio and adiposity (Bernhard et al., 2016; Kyle et al., 2004a; Levitt et al., 2010). These findings highlight the need for caution when applying regression equations beyond the populations in which they were derived. Further work is required on developing and validating reference equations specific to disease states with altered hydration and excess adiposity.

Despite its limitations, there is a growing field of literature on the clinical applications of bioimpedance beyond volume estimations. One primary focus of such research is phase angle, the index of the ratio of reactance to resistance (Institute of Medicine & Committee on Military Nutrition Research, 1997). Phase angle is proposed as a marker for cellular integrity and health and is gaining attention in the prediction of nutritional status and disease outcomes (Rinaldi et al., 2019; Toso et al., 2000).

2.1.2.2 Bioelectrical Impedance Analysis - Multi-frequency

MF-BIA applies the same principles as SF-BIA, except resistance is measured at both low and high frequencies (Kyle et al., 2004a). Low frequencies are unable to overcome the capacitance of the cell membrane and, therefore, are only conducted through the extracellular space (Kyle et al., 2004a). In contrast, high-frequency currents may overcome the 'resistance' of the cell membrane and travel through both the extracellular and intracellular compartments (Kyle et al., 2004a). By applying both low and high-frequency currents, both TBW and the extracellular compartment can be determined. From this, ICW can also be calculated by subtracting ECW from TBW. Thus, unlike in SF-BIA, TBW estimation is not constrained by the assumption of a constant ECW: ICW ratio. Therefore, MF-BIA estimates may be less susceptible to changes in hydration status. The meta-analysis by Martinoli et al. (2003) found SF-BIA and BIS, but not MF-BIA, overestimated TBW in healthy adults, adults with obesity, and in chronic renal failure. The validity of MF-BIA, specifically in populations with obesity, has been reviewed by Becroft et al. (2019), who found that MF-BIA typically overestimates FFM and underestimates FM in these populations. Acceptable populationlevel concordance is observed, but large individual variability. This supports earlier conclusions that MF-BIA methodology lacks sufficient accuracy and reliability to assess body composition at the individual level in states of altered hydration (O'Brien et al., 2002).

2.1.2.3 Bioimpedance Spectroscopy

BIS fundamentally differs from SF-BIA and MF-BIA as it uses mathematical modelling to estimate volumes from resistances obtained across a spectrum of frequencies. A detailed description of the

principles and rationale behind these modelling methodologies is available in a review by De Lorenzo et al. (1997). The primary benefit of BIS is its application in individuals with altered hydration status. Firstly, through the application of very high frequencies, both ICF and ECF can be quantified (Matthie, James R., 2008). Therefore, unlike SF-BIA, assumptions of a normal hydration status are not required. BIS further individualises the estimation of fluid compartments by applying gender-specific resistivity constants derived from dilution experiments (De Lorenzo et al., 1997).

Despite more advanced methodology, validation studies for BIS also typically find large individual variability alongside good population-level agreement (Becroft et al., 2019; Earthman et al., 2007). Additionally, BIS consistently overestimates TBW and FFM in obesity (Cox-Reijven et al., 2002; Mager et al., 2008). Increased margins of error are attributed to the effect of adipose tissue on ICW-specific resistivity (Matthie, James R., 2008). Moissl et al. (2006) attempted to address this by introducing BMI into modelling equations as a marker for adiposity. This approach resulted in significantly more accurate estimations for all subjects, with more drastic effects at statistically extreme BMIs (Moissl et al., 2006). BIS methodology offers clear advantages over traditional BIA techniques, but further modifications to modelling approaches are required to improve its validity and usefulness in a clinical setting.

2.1.3 Direct Methods

2.1.3.1 Isotope Dilution

Isotope dilution is a technique for measuring TBW (and ECF). The underlying principle is that the quantity of tracer administered is the same before and after administration. Dilution procedures take fluid samples, such as blood, urine or saliva, before and after administering tracers such as deuterium (²H) and oxygen-18 (¹⁸O) into the body. TBW can be calculated using the tracer quantity, baseline concentration, and equilibrium concentration. FFM and FM are typically derived by assuming the hydration of FFM is constant at 73% (Pace & Rathbun, 1945; Wang, Z. et al., 2000). However, the proportion of water in FFM may be influenced by disease states, adiposity, and age (Hewitt et al., 1993; Levitt et al., 2010; Waki et al., 1991).

2.1.3.2 Total Body Counting

Total body counting or whole-body counting is a technique that measures high-energy gamma rays from naturally occurring radioactive potassium 40 in the body (Forbes et al., 1991). Potassium is found almost exclusively inside cell bodies and can, therefore, be used to estimate body cell mass (BCM). Using ratios identified in cadaver work from Forbes and colleagues, total body potassium

(TBK) can also be used to estimate FFM (Forbes et al., 1991). The validity of this assumption is challenged by consistent findings of lower mean potassium concentrations within FFM in other studies (Ellis, 2000). Additionally, small differences in TBK/FFM are exacerbated when FFM derived from TBK is used to estimate FM in a two-compartment model (Pierson Jr et al., 1991). This suggests TBK is better applied within multicompartmental models as a measure of BCM alongside other body composition analysis techniques.

2.1.3.3 Neutron Activation

Neutron activation analysis is a technique for measuring the elemental composition of the body (Ellis, 2000). When subjects are exposed to neutron radiation beams, energy is absorbed into nuclei and subsequently released as gamma rays. The energy characteristics and intensity can be used to identify the element and its quantity (with suitable calibration of the methodology). Using this technique, elements like nitrogen, calcium, carbon, and sodium can be measured (Ellis, 2000). Total body nitrogen can subsequently be used to estimate total body protein (Haas et al., 2007; Wang, J. et al., 1993). The distribution and proportions of chemical elements within the body can be used to estimate body composition in a four-compartment model. This is known as elemental partition analysis. Various methods for elemental partition analysis are described in a 1999 review by Kehayias & Valtueña. Neutron activation techniques have been used infrequently due to their high radiation exposure for subjects.

2.1.4 Criterion Methods

1.4.1 Hydrodensitometry

Hydrodensitometry, also known as underwater weighing, was historically considered the gold standard for body composition analysis. Hydrodensitometry involves submerging subjects underwater to estimate their body volume through fluid displacement (Katch et al., 1967). The measurement of body volume can be used with body mass and residual lung volume to calculate body density. Finally, body fat percentage can be derived using equations based on two-compartment models, such as those of Siri (1956) or Brozek (1963).

Air displacement plethysmography (ADP) is a modern alternative that also measures body density. The use of air displacement devices offers some advantages over hydrodensitometry, such as a reduced reliance on subject performance and the need for full immersion under water (Demerath et al., 2002; McCrory et al., 1995). However, both ADP and hydrodensitometry are scarcely used for subjects with obesity as the requirement for lightweight or tight swimwear may discourage and limit participation. Additionally, as FM has a lower density than FFM, subjects with obesity may require

counterweights to remain still or achieve full submersion for underwater weighing (Behnke et al., 1942).

1.4.2 Dual-Energy X-Ray Absorptiometry

DXA is a two-dimensional imaging technique used to estimate total and segmental FM, FFM, and bone mineral density (BMD) (Roubenoff et al., 1993). It is globally accepted as the gold standard for measuring bone mineral density and is considered an accurate measure of FM compared to other techniques (Guglielmi et al., 2016). The fundamental principle behind DXA involves measuring the attenuation of high and low-energy X-rays through the body. The degree of attenuation is dependent on the density and thickness of the tissue it passes through, as well as the energy level of the x-ray (Crabtree et al., 2007). This can be applied to a two-compartment model of the body to estimate bone mineral and soft tissue mass. Bone can be distinguished from soft tissue as its higher density results in greater attenuation (Bazzocchi et al., 2016). This principle can also be applied to a threecompartment model of bone mineral, FM and lean tissue mass. In tissue without bone, the ratio of attenuation between high and low-energy X-rays is found to be proportional to the fat content of the tissue (Laskey, 1996). Manufacturer-specific algorithms are then applied to estimate fat content in tissue where bone is present. By analysing areas of soft tissue with and without bone, FM and lean tissue mass can be differentiated (Toombs et al., 2012).

Like other techniques, DXA relies on assumptions around hydration and tissue density. These assumptions can vary between manufacturers, models and software, which leads to notable variability (Plank, 2005). DXA scans can be completed quickly, in 5-15 minutes and utilise very low doses of radiation (Crabtree et al., 2007). Additional technological developments have improved the practicality of DXA for subjects with obesity, as older devices were not suited for larger body sizes (Tataranni & Ravussin, 1995).

DXA is also less affordable and accessible than other techniques but offers the distinct advantage of enabling regional body composition assessment (Pietrobelli et al., 1996). However, high margins of error have been identified in areas containing minimal bone-free soft tissue, like the arm and thorax (Roubenoff et al., 1993). Validation studies against the gold standard four-compartment model indicate DXA may underestimate FM (Toombs et al., 2012). However, this finding is not consistent between studies. Unlike MRI (magnetic resonance imaging), DXA is unable to distinguish visceral and subcutaneous fat. Software attempts to enable this show strong correlations with CT (computed tomography) measurements but exhibit larger margins of error in subjects with obesity (Meredith-Jones et al., 2018; Taylor et al., 2021). DXA performs well at the group level when tracking changes

in body composition over time, but large intra-individual differences limit its individualised use (Toombs et al., 2012).

1.1.3.3 Magnetic Resonance Imaging

Whole-body MRI is considered the gold standard for body composition analysis as it accurately measures the quantity and distribution of both adipose tissue and skeletal muscle (Mitsiopoulos et al., 1998; Thomas et al., 2013). MRI uses radio waves and strong magnetic fields to generate images of the body's soft tissue (Machann et al., 2005). There are three approaches for assessing body composition in MRI images: single-slice, multi-slice, and whole-body (Hu et al., 2016). Multi-slice and whole-body approaches enable the direct quantification of visceral adipose tissue (VAT), a key component for assessing metabolic disease risk (Thomas et al., 2013).

Unlike other imaging techniques, CT and MRI do not utilise ionizing radiation. Therefore, MRI is preferred for whole-body assessments and in research settings. Despite this, the use of whole-body MRI is infrequent due to costly equipment, availability, and the difficulty of analysing three-dimensional images (Hu et al., 2016). MRI devices were also previously limited by gantry size or table weight capacity. However, the designs of newer imaging devices address these issues and make MRI more accessible for populations with obesity (Rothschild et al., 1991).

2.1.5 Comparison of DXA and Bioimpedance Techniques

DXA is traditionally accepted as the preferred method for body composition analysis due to its validation against the gold standard four-compartment model (Toombs et al., 2012). The complexity of four-compartment models makes them tedious and impractical for use in clinical settings or large studies. The affordability and accessibility of bioimpedance instruments provide an alluring alternative to DXA in these settings. However, there is limited evidence validating bioimpedance techniques against four-compartment models or DXA. Additionally, determining the validity of bioimpedance techniques is further complicated by variations between devices, equations, and measurement techniques. The high heterogeneity between studies often means meta-analysis is not possible. This limitation is exacerbated in clinical subpopulations like bariatric surgery patients as the available literature is further reduced. A summary of studies comparing body composition measures from BIA and DXA can be found in Table 3.

Conclusions on the validity of bioimpedance techniques must carefully consider the BMI of study populations. In a 2018 retrospective study of 3,655 individuals, Achamrah et al. compared body composition measures from DXA and SF-BIA at 50 Hz. Strong correlations between DXA and SF-BIA for both FM and FFM were observed at the population level alongside large limits of

agreement. For BMI \geq 18.5 kg/m² and BMI < 40 kg/m², compared to DXA, SF-BIA overestimated FFM by 3.38-8.28 kg and underestimated FM by 2.51-5.67 kg. Interestingly, SF-BIA overestimated FFM by 5.87 kg in BMI \geq 40 and underestimated FFM by 2.25 kg in BMI < 16 kg/m². Differences in FM and FFM between SF-BIA and DXA increased with BMI in patients between 18.5 and 25 kg/m², and \geq 40 kg/m² (Achamrah et al., 2018). Two key conclusions can be drawn from these findings.

- For BMI ≥ 18.5 kg/m² SF-BIA overestimates FFM and underestimates FM compared to DXA.
- 2) The validity of FM and FFM determination by SF-BIA is poor at the individual level.

These findings are in agreeance with other comparative studies of SF-BIA and DXA (Savastano et al., 2009; Widen et al., 2014). Findings from Savastano et al. are notable as measures of FM and FFM were compared at baseline, six months and twelve months following LAGB. The overestimation of FFM by SF-BIA was diminished at both follow-up time points and was attributed to the participants' weight loss. Investigations into MF-BIA have produced identical findings where FFM is overestimated, and FM is underestimated in participants classified as overweight and/or obese. Six studies identified in a systematic review by Becroft et al. (2019) reported that the inclusion of algorithmic adjustments for BMI and waist circumference improved the accuracy of FM and FFM by MF-BIA. However, no detailed information was provided on the adjustments made.

Author (Study Design)	Study Population	BMI $(kg/m^2)^1$	BCA Techniques	Bioimpedance methods compared to reference method:		Comments
				FFM	FM	
Achamrah et al., 2018 (Retrospective, Cohort)	Nutrition Unit Patients seen for malnutrition, obesity, or eating disorder. (n=3,655)	Analysis split into BMI ranges: < 18.5 (n=379) 18.5 - 25 (n=237) 25 - 30 (n=328) ≥ 30 (n=2,711)	MF-BIA & Whole body DXA	FFM strongly correlated irrespective of BMI (r=0.89, p<0.0001). In BMI <16 kg/m ² underestimated FFM by 2.2kg. Overestimated FFM in 18.5≤BMI<25 kg/m ² by 3.3 kg and in 25≤BMI<30 kg/m ² by 7.1 kg.	FM strongly correlated irrespective of BMI (r=0.95, p<0.0001). In BMI <16 kg/m ² , overestimated FM by 2.5kg. Underestimated FM in 18.5≤BMI<25 kg/m ² by 2.5 kg and in 25≤BMI<30 kg/m ² by 5.6 kg.	MF-BIA device used only used 50kHz to calculate TBW. Large limits of agreement (LOA) were observed in all BMI groups. In patients with normal BMI or BMI ≤40 kg/m ² , differences between FFM and FM increased with BMI.
Ballesteros- Pomar et al., 2022 (Cross-sectional)	Severe obesity. (n=115)	46.1	MF-BIA & Whole body & segmental DXA	Determined by a proprietary equation, overestimated ALM by 7.3 kg. Equations by Sergi, Kyle and Yamada overestimated ALM by 2.8 kg, 4.1 kg, & 2.7 kg respectively.	Determined by proprietary equation, underestimated FM by 3.4 kg.	Large LOA observed for FM (14 kg). Across equations, larger LOA and bias compared to DXA was typically found in male subjects for ALM.
Berstad et al., 2012 (Prospective, Cohort)	Randomly sampled participants in a Norwegian sleep apnoea study. (n=93)	30.9	BIS & Whole body DXA	Overestimated FFM by 3.5 kg.	Underestimated FM by 3.5 kg. Underestimated BF% by 3.8%.	

Table 3. Summary of Studies comparing body composition measured by bioimpedance techniques to reference methods.

Faria et al., 2014 (Cross-sectional)	Patients undergoing pre-operative treatment for bariatric surgery. (n=73)	40.2	MF-BIA & Whole body DXA	Overestimated FFM by 1.28 kg.	Underestimated FM by 2.05 kg.	Participants with body weights exceeding 120 kg were excluded due to the DXA device capacity. 89% of participants were female.
Savastano et al., 2009 (Prospective, Cohort)	Obese Women undergoing LAGB. (n=45)	42.1 at baseline.	SF-BIA @ 50 Hz & Whole Body DXA	FFM strongly correlated at T0 ($r^2=0.87$), T6 ($r^2=0.82$), & T12 ($r^2=0.99$).	FM strongly correlated at T0 ($r^{2}=0.98$), T6 ($r^{2}=0.94$), & T12 ($r^{2}=0.99$). BF% strongly correlated at T0 ($r^{2}=0.91$), T6 ($r^{2}=0.89$), & T12 ($r^{2}=0.98$).	Correlation between SF-BIA & DXA for FM & FFM increased with weight loss.
Verdich et al., 2011 (Prospective, Cohort)	Obese women participating in the NUGENOB study. (n=131)	33.8 & 35.7. No mean BMI reported for the entire cohort.	MF-BIA & Whole body DXA (Two devices)	Overestimated FFM by 3.04 kg and 1.95 kg at baseline (p <0.001). Overestimated Δ FFM by 0.47 kg (p <0.05).	Underestimated FM by 3.17 kg and 0.98 kg at baseline (p<0.001 & p<0.05). Underestimated ΔFM by 0.38 kg (N.S)	Strong correlations were found between DXA and MF-BIA for FM & FFM at baseline and following weight loss. Wide LOA observed.
Widen et al., 2014 (Prospective, Cohort)	Bariatric Patients. (n=50)	44.1 at baseline	SF-BIA & a 3- compartment model consisting of D ₂ 0 dilution & Bod Pod.		BF% is highly correlated with the three-compartment model BF% at T0, T12, and weight loss. Underestimated BF% at T0 & T12 by 3.3% (p<0.001) & 1.66% (p=0.03).	SF-BIA used foot-to-foot electrodes. Large LOA observed for BF% (T0 19.3%, T12 19.1%, and total change 20.4%)

¹Mean BMI of total study cohort unless stated otherwise. Abbreviations: ALM = Appendicular Lean Mass; BCA = Body Composition Analysis; BF% = Body fat percentage; BIS = Bioimpedance spectroscopy; BMI = Body mass index; DXA = Dual-energy x-ray absorptiometry; FM = Fat mass; FFM = Fat-free mass; LOA = Limits of agreement; MF-BIA = multi-frequency bioelectrical impedance analysis; SF-BIA = Single-frequency bioelectrical impedance analysis.

2.2 Bariatric Surgery and Body Composition

2.2.1 Bariatric Surgery

The primary mechanisms of weight loss following bariatric surgery are malabsorption and restriction. Malabsorptive procedures like the jejunoileal bypass limit intestinal nutrient and caloric absorption through shortening or bypassing lengths of the small intestine. Alternatively, restrictive procedures such as gastric banding or the SG reduce caloric intake by invoking early satiety through reductions in stomach capacity. The RYGB and biliopancreatic diversion are examples of procedures with both restrictive and malabsorptive outcomes. The SG has replaced the gastric bypass as the most frequently performed procedure in New Zealand and internationally despite there being no clear consensus on procedure superiority (Garrett et al., 2020).

2.2.2 Sleeve Gastrectomy

The SG involves the resection of the stomach through stapling along an orogastric bougie. Approximately 75% of the stomach is removed along the greater curvature, with the resected stomach resembling a tubular sleeve (Huang et al., 2019). The SG was initially utilised for weight loss in 'super-obese' patients prior to a secondary procedure, usually a gastric bypass or biliopancreatic diversion with duodenal switch (BPD-DS) (Marceau et al., 1998). However, over time, SG has grown in popularity as a standalone procedure due to the procedure's relative ease of technique, low complication risk and low mortality rates (Gagner et al., 2013; Udelsman et al., 2019).

The alteration to the stomach's structure influences body weight and metabolic function through numerous factors. Being a restrictive technique, the primary weight loss mechanism of the SG is reduced oral food intake. Food intake is reduced following an SG due to reductions in stomach capacity and reductions in the appetite-stimulating hormone ghrelin (Huang et al., 2019). Ghrelin is produced in parietal cells found in the fundus and body of the stomach and previous studies have observed significant reductions in fasting ghrelin levels and suppression of post-prandial ghrelin production five years following SG (Bohdjalian et al., 2010; Karamanakos et al., 2008).

2.1.2 Roux-en-Y Gastric Bypass

The RYGB also involves segmentation of the stomach into an upper and lower portion through stapling. The upper portion of the stomach or gastric pouch acts similarly to the SG to reduce stomach capacity and limit food intake (Rubino et al., 2010). The gastric pouch is anastomosed to the jejunum, resulting in a bypassing of the distal stomach, duodenum, and proximal jejunum. This is known as the Roux-limb and typically measures 100 cm in length. The proximal jejunum is then anastomosed to the distal jejunum, which is known as the biliopancreatic limb. This results in food

bypassing the distal stomach, duodenum, and proximal jejunum (Rubino et al., 2010). The bypass reduces the capacity for nutrient absorption, further contributing to weight loss (Abdeen & Le Roux, 2015). Long-term considerations of RYGB include the management of nutrient deficiency risk and subsequent complications such as anaemia and osteoporosis (Abdeen & Le Roux, 2015).

2.2.3 Weight Loss

The effectiveness of bariatric surgery in producing weight loss is consistently upheld in the literature. A systematic review and meta-analysis of RCTs conducted in 2016 by Cheng et al. reported that surgical interventions resulted in greater weight loss than non-surgical approaches in RCTs with 1-year, 2-year and long-term (3-year or greater) follow-up durations (Cheng et al., 2016). The superiority of surgical procedures was maintained irrespective of surgery type, with greater weight loss observed following SG, RYGB, BPD and laparoscopic adjustable gastric banding (LAGB).

Weight loss after bariatric surgery is highly variable and influenced by factors such as the presence of comorbidities, pre-operative weight, and post-operative behaviours such as physical activity and dietary habits (Livhits et al., 2011). Estimating the degree of weight loss following bariatric surgery is limited by the lack of studies featuring follow-up durations greater than three years. In a 2021 review by van Rijswijk et al., the mean percentage total weight loss (%TWL) ranged from 27.5% to 32.3% for RYGB and 23.9% to 26.1% for SG (Van Rijswijk et al., 2021). The superiority of any specific procedure is currently not supported. However, currently, available comparative evidence of weight loss outcomes achieved from RYGB and SG favours RYGB. A 2021 meta-analysis by Uhe et al. included 20 studies comparing RYGB and SG and found RYGB resulted in greater %TWL at three months, six months, one year, two years, and five years post-surgery. RYGB also produced a greater percentage of effective weight loss (%EWL) than SG at 3-year and 5-year follow-ups (Uhe et al., 2022). However, it is worth noting that the use of non-standardised definitions of ideal body weight (IBW) and preoperative weight can generate significant variation in %EWL (Montero et al., 2011).

2.2.4 Fat Mass

Correlations between excessive adiposity, inflammation and ill health have been made for decades (Baumgartner et al., 1995; Vahdat et al., 2012). Clinical advancements in reducing metabolic disease risk have and will continue to target reductions in body FM alongside the preservation of FFM. Developments in bariatric surgery trends are guided by these two objectives.

Despite being referred to as weight loss surgery, the primary purpose of bariatric procedures is the reduction of body FM. The effect of SG and RYGB on changes in FM within the first year following surgery has been less extensively studied than total body mass (Table 4). Additionally, significant variability exists between studies by surgical specifications (such as SG bougie volume and RYGB limb length), methods of body composition analysis, and reported measures of body composition.

One prospective, non-randomised study by Otto et al. (2015) included 173 individuals and found that the group receiving RYGB had a lower body fat percentage (BF%) than SG after one year. However, the RYGB group had a lower preoperative BF% and no statistically significant difference was found when adjusted for BMI. Further studies by Kavanagh et al. (2019), Keidar et al. (2013), and Wells et al. (2015) also found no difference between the percentage change in FM between SG and RYGB. Notably, bioimpedance techniques or ADP were used to measure body composition within these studies. Comparisons utilising DXA to measure body composition are further limited. Guerrero-Perez et al. (2019) conducted a randomised study but only compared changes in FM, not BF% or percentage change in FM. Moize et al. (2013) also used whole-body DXA in a non-randomised study of 50 individuals and found no significant change in BF% one year post-surgery between RYGB and SG.

While high variability is observed in the extent of FM reduction twelve months following SG or RYGB, both procedures appear to produce similar reductions in FM percentage ranging from 10-25%. Like weight loss, FM reduction occurs continuously over the first twelve months following bariatric surgery, with the largest reductions occurring during the first 3-6 months. Trends in FM reduction beyond 12 months have been rarely investigated. One 2021 study by Ceriani et al. followed bariatric patients over five years, including 48 undergoing RYGB and 46 undergoing SG, with mean changes in FM being 33.6 kg and 26.6 kg, respectively. However, the study aimed to compare BPD and long common limb revision BPD to RYGB and SG. Thus, no statistical comparisons were made between the RYGB and SG groups. Further investigations into the effects of SG and RYGB on FM are required before drawing conclusions with any certainty.

2.2.5 Fat Mass Distribution

In addition to total body FM, the effect of bariatric surgery on FM distribution is of notable clinical importance. Investigations into body fat distribution have repeatedly found that high visceral FM contributes to increased metabolic disease risk independent of total FM or body weight (Kissebah et al., 1982; Lopes et al., 2016; Stefan, 2020). Most data available utilises single abdominal slices from MRI techniques to estimate VAT and subcutaneous adipose tissue (Korner et al., 2008). Few

studies have directly quantified the entirety of the sub-depots of adipose tissue before and after bariatric surgery.

Toro-Ramos et al. (2015) utilised whole-body MRI to quantify changes in FM and its distribution in the first 24 months following bariatric surgery. Significant reductions in total adipose tissue (TAT), subcutaneous adipose tissue (SAT), intramuscular adipose tissue (IMAT) and VAT occurred during the first 12 months following surgery (Toro-Ramos et al., 2015). No further significant changes were observed between 12- and 24 months post-surgery except in females. Reductions in both VAT and IMAT occurred despite no further significant reductions in body weight. Additionally, both male and female subcategories in the surgical group had similar VAT values to non-surgical controls after 24 months despite significantly higher body weight, TAT & SAT post-surgery (Toro-Ramos et al., 2015). This suggests that bariatric surgery may have targeted effects on VAT. This finding is supported by those of Schneider et al. (2016), where greater reductions in truncal fat were observed compared to leg fat in 42 individuals following LRYGB or SG.

Comparative investigations between surgery types are further scarce. Schneider et al. (2016) also compared changes in fat distribution between the two surgery types and found no differences in truncal or leg fat reduction. This contrasts with findings from Kayshap et al. (2013) where a greater absolute reduction in percent truncal fat was observed in RYGB compared to SG (16% vs 10% p=0.04) despite similar total weight loss. The utilisation of whole-body DXA is a key limitation of both studies, as adipose tissue sub-depots could not be quantified. One study by Kennegott et al. (2019) compared changes in adipose tissue quantified by whole-body MRI and found no differences in relative changes in VAT or SAT 12 months following RYGB and SG.

2.2.6 Fat-Free Mass

Lean mass or FFM losses typically comprise 20-30% of total weight loss in individuals classified as overweight or obese (Bradley et al., 2012; Magkos et al., 2016; Santanasto et al., 2011). Skeletal muscle represents the largest component of FFM, and in addition to generating contractile forces for movement, it plays important roles in metabolic function. Skeletal muscle is the primary site for insulin-mediated glucose absorption and largely determines resting energy expenditure (Baron et al., 1988; Zurlo et al., 1990). Low levels of skeletal muscle mass (SMM) are associated with impaired glucose homeostasis, osteoporosis, and increased mortality risk (Jang et al., 2020; Park & Yoon, 2013; Wang, Y. et al., 2023). Thus, the contribution of FFM to weight loss is a critical consideration when analysing weight loss interventions.

Trends in lean body mass (LBM), FFM, and SMM following bariatric surgery have been well documented. Strong correlations are observed between total weight loss and losses of LBM, FFM & SMM (Nuijten et al., 2022). Similar to FM, greater rates of lean tissue loss occur during the first three months, with more gradual losses occurring up to twelve months post-surgery. Pooled findings from a meta-analysis by Nujiten et al. (2022) found that 55% of LBM losses, 52% of FFM losses and 66% of SMM losses observed at twelve months occurred during the first three months following bariatric surgery. Additionally, of the eight studies identified that assessed LBM at 12 months and between 18-36 months, a mean loss of only 1.29 kg was found (Nuijten et al., 2022). This shows that lean tissue mass remains relatively stable beyond 12 months.

Comparisons of the effect on lean tissue of different types of bariatric surgery have rarely identified differences. In the previously mentioned meta-analysis by Nujiten et al. (2022), no differences were identified between procedures when comparing LBM, FFM and SMM relative to total weight loss. Comparative studies inconsistently report superior preservation of absolute LM following RYGB compared to SG (Davidson et al., 2018; Otto et al., 2015; Wells et al., 2015). However, in concordance with conclusions from pooled data, these effects are abolished when adjusting for differences in total weight loss or body composition at baseline.

Author (Study Design)	BCA technique	Sample size n(%F)	Variable	RYGB	SG	Comments	
		· · · ·	%TWL	35.3%	27.3%	The study was not powered to detect differences in	
Guerrero-Perez et al., 2020	Whole-body	30 (60)	ΔFM	-19.6 kg	-11.6 kg	FM or FFM between surgery groups.	
(Prospective, randomised)	DAA		ΔLM	-9.6 kg	-6.1 kg		
			TWL	27.9 kg	27.3 kg	No significant differences were identified between the	
			%EWL	53.4%	47.2%	surgery groups.	
Kavanagh et al., 2019 (Prospective, non- randomized)	ADP	66	ΔFM	-23.3 kg (10.5%)	-23.3 kg (9.2%)		
(anaonizea)			ΔLM	-4.6 kg (10.5%)	-3.9 kg (9.4%)		
Keidar et al., 2013 (Prospective, randomised)	BIA			%TWL	25.9%	28.4%	No significant differences were identified between the
		BIA 37 (57)	ΔFM	-23.9 kg (23.7%)	-24.9 kg (24.2%)	surgery groups.	
			ΔFFM	-5.3 kg (6.9%)	-9.3 kg (9.1%)		
	Whole-body DXA		%TWL	32.1%	33.5%	No significant differences in weight loss or fat loss.	
Moize et al., 2013			ΔFM	-31.3 kg	-31.9 kg	%LM loss was greater in SG than RYGB after adjusting for baseline LM.	
(Prospective, non- randomized)		50 (82)	$\Delta BF\%$	-15.2%	-13.5%		
Tundonnized)			ΔLM	-6.2 kg	-9.3 kg		
			%TWL	31.7%	30.5%	No significant differences were found between the two	
Otto et al., 2015 (Prospective, non-randomized)		BIA 173 (71)	%EWL	62.9%	52.3%	groups after adjusting for baseline BMI. At baseline mean BMI was 45.6 kg/m ² in RYGB (n=127) and 55.9 kg/m ² in SG (n=46)	
	BIA		ΔFM	-30.2 kg	-37.6 kg		
			$\Delta BF\%$	-15.5%	1-2.1%		
				ΔLM	-9.7 kg	-14.7 kg	

Table 4. Summary of studies comparing body composition outcomes 1-year after SG and RYGB

			%EWL	70.4%	32.1%	Analysed data only included follow-up data from <12
Strain et al., 2009			$\Delta BF\%$	-20.9%	-17.7%	months after surgery. Mean follow-up times were 19.1 \pm 10.6 months in RVGB and 16.7 \pm 5.6 months in SG
(Prospective, non- randomized)	BIA	131 (73)	ΔLM	16.0%	9.1%	Few direct comparisons were made between the two groups. The analysis included ANCOVA for four
						groups: RYGB, SG, BPD-DS & AGB.
 			ΔFM	-39.8 kg	-37.4 kg	Data was originally published in lbs.
Wells et al., 2015	DIA	52 (07)	$\Delta BF\%$	-7.7%	-4.9%	A Baseline difference in BMI between groups was reported.
(Retrospective, non- BIA randomized)	53 (87)	ΔFFM	- 6.3kg	-6.6 kg	A larger increase in FFM% was observed in RYGB than in SG.	
			$\Delta FFM\%$	17.1%	10.8%	

Reported values represent the reported group means. Sample sizes are expressed as n (Percentage of sample who are female). Sample sizes only include study participants allocated to RYGB or SG groups. Other surgical groups are not included in the reported value.

Abbreviations: ADP = Air displacement plethysmography; AGB = Adjustable gastric banding; BCA = Body Composition Analysis; BF% = Body fat percentage; BMI = Body mass index; BPD-DS = Biliopancreatic diversion with duodenal switch; <math>DXA = Dual-energy x-ray absorptiometry; EWL = Excessive weight loss; FM = Fat mass; FFM = Fat-free mass; LM = Lean Mass; RYGB = Roux-en-Y gastric bypass; SG = Sleeve gastrectomy; TWL = Total weight loss.

Author (Study Design)	BCA technique	Sample size n(%F)	Variable	RYGB	SG	Comments
Ceriani et al., 2021			%EWL	81.7%	78.6%	No significant differences were found between the groups at 5-years.
(Retrospective, non-	BIA	94 (85)	ΔFM	-33.6 kg	-26.6 kg	The retrospective analysis included RYGB patients since 2003 and SG patients since 2010. The study
randomized)			ΔFFM	-11.2 kg	-9.8 kg	kg since 2003 and SG patients since 2010. The study included BPD (2002) and BPD+LCL-R (2007) groups.
Davidson et al., 2018 (Prospective, non- randomized)			TWL	41.8 kg	40.0 kg	The sample size was not consistent between the groups (RYGB: 58; SG: 11). An additional 14 males underwent RYGB. However, pooled data was not
	3-Compartment model using Bod Pod & D ₂ 0 dilution. Whole-body MRI	69 (100)	ΔFFM	-10.2 kg	-6.2 kg	published.
						BMI was higher, and FFM% was lower in the SG group than in RYGB at all time points.
			∆FFM%	12.9%	8.7%	5-year data was only available in 62% of participants.
			%TWL	31.0%	27.1%	A non-significant difference in baseline weight was present between the groups (p=0.055). However,
Yang et al., 2015 (Prospective, randomised)	sed) Only body weight.	64 (66)	TWL	29.5 kg	24.3 kg	height was greater in the RYGB group, and no difference in baseline BMI was detected.
_ ,			%EWL	92.3%	81.9%	At three years, %TWL, TWL and %EWL were significantly greater in RYGB than SG.
Zhang et al., 2014 (Prospective, randomised)	Only body weight.	64 (59)	%EWL	76.2%	63.2%	%EWL was significantly higher in RYGB than SG at 2-years, 3-years, 4-years, and 5-years.

Table 5. Summary of studies comparing body composition outcomes \geq 3-years after SG and RYGB

Reported values represent the reported group means. Sample sizes are expressed as n (Percentage of sample who are female).

Abbreviations: BCA = Body Composition Analysis; BMI = Body mass index; BPD = Biliopancreatic diversion; BPD+LCL-R = Revisional long common limb biliopancreatic diversion; EWL = Excessive weight loss; FM = Fat mass; FFM = Fat-free mass; RYGB = Roux-en-Y gastric bypass; SG = Sleeve gastrectomy; TWL = Total weight loss.

2.2.7 Type 2 Diabetes Mellitus Remission

Body composition changes following bariatric surgery also have significant effects on metabolic health. Pooled data from a systematic review of RCTs found five-year T2DM resolution rates (HbA_{1C} below 6.0%) of 37.4% following RYGB and 27.5% following SG (Sharples & Mahawar, 2019). However, differences in T2DM resolution rates, improvement rates or HbA_{1C} levels at five years failed to reach statistical significance. While not statistically significant, differences in T2DM resolution rates may be explained by the greater %EWL achieved following RYGB in the included studies.

In a study of 60 subjects with uncontrolled T2DM, those receiving RYGB achieved greater weight loss and had significantly lower HbA_{1C} levels at 24 months post-surgery than those receiving SG or intensive medical therapy (Guerrero-Pérez et al., 2019). Additionally, 58% of SG patients and 80% of RYGB patients achieved T2DM remission (defined as an HbA_{1C} below 7.0% or 53 mmol/mol). Kashyap et al. (2013) also reported superior outcomes following RYGB compared to SG. Participants randomised to undergo RYGB had improvements in insulin sensitivity and beta cell function compared to the control group receiving intensive medical treatment (Kashyap et al., 2013). No improvements were found in the SG group. These differences were attributed to the greater reduction in truncal fat in the RYGB group. In contrast, Yang et al. (2015) found no difference in HbA_{1C} values three years after SG or RYGB despite greater weight loss occurring in the RYGB group. Interestingly, a greater reduction in HbA1c values was observed at 3 and 6 months in the RYGB group.

Analysis of T2DM remission outcomes at five years of 114 individuals from this trial cohort has been previously published (Murphy et al., 2022). Participants were randomly assigned to undergo RYGB or SG and were assessed at one year and five years post-operatively. T2DM remission was defined as an HbA_{1C} <6% or 42 mmol/mol. T2DM occurred at a greater rate following RYGB (47%) than SG (33%). Percentage weight loss was also greater in the RYGB group. Applying the definition of T2DM remission used by Guerrero-Pérez (2019) (HbA_{1C} <7.0%) produces similar remission rates of 70% in RYGB and 58% in SG (Murphy et al., 2022).

Currently, there is insufficient evidence to claim superior T2DM remission outcomes following one type of bariatric surgery. Despite this, potential physiological explanations are well established. Improvements in insulin sensitivity following bariatric surgery are primarily attributed to reductions in body weight and body FM. Further, markers of obesity, such as abdominal adiposity, have long been correlated with insulin sensitivity (Goodpaster et al., 1997). With the previously discussed evidence suggesting quantitative and qualitative differences in body composition outcomes

following bariatric surgery, further investigation into T2DM remission and diabetes-associated factors is warranted.

Chapter 3. Methods

3.1 Randomised Clinical Trial

3.1.1 Ethics and Trial Registration

This clinical trial (Murphy et al., 2016) is registered with the Australian New Zealand Clinical Trials Registry (ACTRN12611000751976). Ethics approval was obtained from the Northern Y Regional Ethics Committee (NTY/11/07/082).

3.1.2 Power Analysis

Power analysis was completed for the primary outcome of the study, T2DM remission. A minimum of 42 patients per arm was calculated to provide 80% power to detect a difference between the two groups using a two-sided alpha value of 0.05. After adjusting for an expected loss to follow-up rate of 20%, a minimum of 53 patients per arm was required (Murphy et al., 2022).

3.1.3 Patient eligibility and recruitment

Patients referred to the bariatric surgery program at North Shore Hospital aged 20-55 years, with a BMI of 35-65 kg/m² and a diagnosis of T2DM at least six months prior, were invited to attend an information evening for the study. Eligible participants were also required to be suitable for either surgical procedure and able to commit to long-term follow-up. Exclusion criteria included pregnancy, type 1 diabetes mellitus, chronic pancreatitis, oral steroid therapy, active smokers, previous bariatric or esophagogastric surgery, myocardial infarction or cerebrovascular events within the last six months, malignancy in the last five years and those with contraindications for general anaesthesia. Recruitment of participants began in September 2011 and was completed in October 2014. A total of 114 participants were recruited from a single centre (North Shore Hospital).

3.1.4 Study Design

The study was a single-centre, prospective, parallel, two-arm, randomised, double-blind, superiority trial. The trial protocol has been previously published (Murphy et al., 2016). The primary outcome assessed was the comparative rates of T2DM remission following RYGB or SG. Secondary outcomes included weight change, the resolution of comorbidities, peri- and post-operative complications, change in body composition, resting energy expenditure and bone density, and

quality of life. A sub-study was completed with additional outcomes related to T2DM remission, including food intake and satiety, as well as changes in gut hormones and gut bacteria.

Participants providing informed consent and with confirmed eligibility were placed on a very lowcalorie diet (VLCD) two weeks prior to surgery. Two days prior to surgery, participants attended a pre-surgery appointment at the body composition laboratory at Auckland City Hospital. Baseline data obtained included height, weight, BMI, DXA, SF-BIA, MF-BIA and BIS measurements. Participants were invited to attend an identical appointment at one- and five-years post-surgery, where measurements were collected again. The methods used during these assessments are described in detail in section 3.3.

Participants were randomised to a surgery type on the day of surgery. Both the assessor and patient were blinded to the surgery they had received. All surgeries were performed under the direct supervision of an experienced bariatric surgeon at North Shore Hospital, Auckland, New Zealand. Further details on the procedures are provided in section 3.2.3.

3.2 Study Protocol

3.2.1 Very Low-Calorie Diet (VLCD)

All participants were prescribed a VLCD for two weeks preoperatively. This consisted of three daily servings of OPTIFAST (Nestlé, Vevey, Switzerland), each containing ~150 kCal plus vegetables.

3.2.2 Randomisation and Blinding

Participants were randomised to either LSG or SR-LRYGB through computer-generated random number codes (Minim, London) managed by an independent study member and included minimisation based on ethnicity (Māori, Pacific, NZ European/other), BMI category (35-44.9 kg/m², 45-54.9 kg/m² or 55-65 kg/m²), age category (20-29, 30-39 or 40-55), duration of T2DM diagnosis (<5, 5-10 or >10 years), and the presence of insulin therapy.

Participant allocation was disclosed to the surgical team following the administration of general anaesthesia to the participant. Surgery allocation was not disclosed to participants, researchers or other clinicians throughout the study.

3.2.3 Surgical Procedures

Identical laparoscopic incisions were used during both surgical procedures to maintain blinding of participant surgery allocation.

3.2.3.1 Laparoscopic Sleeve Gastrectomy

Participants receiving the laparoscopic SG had a sleeve fashioned using a laparoscopic stapling device over a 36-Fr oro-gastric bougie starting 2 cm proximal to the pylorus.

3.2.3.2 Laparoscopic Silastic Ring Roux-en-Y Gastric Bypass

Participants receiving the laparoscopic RYGB had a lesser-curve based gastric pouch fashioned over a 32-Fr oro-gastric bougie, with a 100-cm antecolic Roux limb anastomosed by a hand-sewn single-layer gastrojejunostomy. Biliopancreatic limbs measured 50 cm. A 6.5 cm silastic ring was placed 2-3 cm above the gastrojejunostomy to prevent stomal dilation.

3.2.4 Post-Operative Management

Both groups received identical postoperative management and follow-up. All pharmacological agents for diabetes, hypertension and hyperlipidaemia were stopped at the time of surgery. Participants were reviewed by an endocrinologist at six weeks, nine months and annually for medication adjustments and monitoring of vascular complications. The endocrinologists remained blinded to which surgery participants had received. Glucose-lowering agents were restarted in participants whose postoperative capillary glucose exceeded 12 mmol/L. The protocol for medication adjustment was published with the study protocol (Murphy et al., 2022).

3.3 Outcome Measures

3.3.1 Anthropometry

3.3.1.1 Body Weight

Body weight was assessed two days prior to surgery following two weeks of VLCD and at one year and five years post-operatively. Weight was measured to the nearest 0.1 kg using digital scales (SECA, Chino, California, USA) at the University of Auckland's Body Composition Laboratory, Auckland City Hospital. The removal of shoes and any heavy clothing items was requested. Adjustment was made for estimated weight of clothing.

3.3.1.2 Height

Height was recorded to the nearest 0.5 cm at each of the three clinical assessments using a stadiometer. Without shoes, participants were asked to stand with their backs and heels against the stadiometer, with their heads in the Frankfort horizontal plane.
3.3.1.3 Anthropometric Derivatives

BMI was calculated for each time point as the participant's weight in kilograms divided by their height in meters squared.

$$BMI (kg/m^2) = \frac{Body Weight}{Height^2}$$

%EWL standardises weight loss against an 'ideal' body weight (IBW) at a BMI of 25 kg/m². %EWL was calculated at one year and five years following surgery using the following formula:

$$\% EWL = \frac{(Weight_{Baseline} - Weight_{Follow\,up})}{(Weight_{Baseline} - IBW)} x\ 100$$

3.3.2 Dual-energy X-ray Absorptiometry

DXA (iDXA, software V.15, GE-Lunar, Madison, Wisconsin, USA) was used to measure total FM, total lean soft tissue mass, total bone mineral content, and fat, lean soft tissue and bone mineral content of the limbs. ASMM was derived according to Heymsfield et al. (1990):

ASMM = limb lean soft tissue mass - 0.82 x limb bone mineral content

Total FFM was calculated as the sum of total lean mass and total bone mineral content. Regional analysis provided measures of abdominal fat and abdominal visceral fat, with the latter estimated using CoreScan[™] software. Subcutaneous abdominal fat was calculated by subtracting abdominal visceral fat from total abdominal fat. Scans were performed with participants in the supine position. Where participants could not fit entirely within the scan area, the composition of the left arm was assumed to be that of the fully scanned right arm.

3.3.3 Bioimpedance Assessment

SF-BIA & MF-BIA measurements were carried out in the supine position immediately following the DXA scan.

3.3.3.1 Single-Frequency Bioimpedance Analysis

A four-terminal impedance analyser operating at 50 kHz and 800 mA was used to measure wholebody resistance (R) and reactance (Xc) following the manufacturer's instructions (Model BIA-101, RJL Systems, Clinton Township, MI, USA). Gel electrodes (#0525, Bodystat Ltd, UK) were placed on the dorsal surfaces of the right hand and right foot, at the distal metacarpals and metatarsals, respectively, and between distal prominences of the radius and the ulna at the wrist, and the medial and lateral malleoli at the ankle. The measured values of R and Xc were entered into a computer program supplied by RJL systems, along with the participant's weight, height, age and sex, to provide estimates of TBF. FFM was calculated by subtracting TBF from body weight.

3.3.3.2 Multi-frequency Bioimpedance Analysis

R and Xc at 50 kHz and impedance values at 5, 50, 100 and 200 kHz were measured immediately following the single-frequency measurements using the four-terminal Quadscan 4000 (Bodystat Ltd, UK) multi-frequency analyser operating at 800 mA. The manufacturer's proprietary equations were used to obtain estimates of TBF based on the participant's weight, height, age, and sex. FFM was calculated by subtracting TBF from body weight. Data are presented as mean \pm standard error of the mean unless stated otherwise.

3.4 Statistical Analysis

Repeated measures data were analysed using linear mixed models to compare changes over time between the groups. A significant group x time (operation x time) interaction effect indicated that the groups differed in their responses over time. A non-significant interaction indicated that the time profiles were similar for the two groups, i.e., the responses over time were essentially parallel to one another. Comparisons between groups for differences in variables of interest were conducted using Fisher's exact test for categorical data, Student's two-sample t-tests for normally distributed data and paired t-tests for within-group changes. Comparison results obtained by DXA and SF-BIA/MF-BIA equations were compared using scatter plots and the Bland-Altman approach (Bland & Altman, 1986), providing the bias with 95% confidence limits (limits of agreement) and the correlation coefficient indicated the extent of proportional bias. Pearson correlations and the concordance coefficient of Lin (1989) were determined for the scatter plots between DXA and BIA data. For FFM and FM, the accuracy of each BIA equation was calculated by the percentage of patients with a difference between BIA and DXA results within $\pm 10\%$ or $\pm 5\%$ of the data obtained by DXA. Statistical analysis was performed using Excel (Microsoft, Redmond, WA) or SAS release 9.4 (SAS Institute, Cary, NC). A p-value <0.05 was considered statistically significant.

Chapter 4. Results

4.1 Patient Characteristics

Body composition data were available at all three follow-up points for 91 of the 114 patients measured at baseline. Exclusion rates were similar between the surgery groups, with 10 RYGB (18%) and 13 SG (22%) patients being lost to follow-up, for a total of 20%. A further three patients were excluded from bioimpedance analysis due to missing either single-frequency or multi-frequency measurements at baseline. The demographics of participants included in the data analysis are shown in Table 6. The mean BMI of the SG group (40 kg/m²) was slightly higher than in the RYGB group (38.9 kg/m²). However, this was not statistically significant (p=0.37). DXA measurements of baseline body composition of the two groups are shown in Table 7. There were no significant differences in FM or fat-free variables between the groups.

Parameter	RYGB (n=46)	SG (n=45)	P value
Age, years	46.8 ± 6.7	47.3 ± 6.1	0.74
Weight, kg	113.6 ± 19.5	115.4 ± 21.3	0.67
BMI, kg/m ²	38.9 ± 5.2	40.0 ± 6.0	0.37
Sex, n (%)			0.68
Male	20 (43.5)	22 (48.9)	
Female	26 (56.5)	23 (51.1)	
Ethnicity, n (%)			0.38
NZ European	27 (58.7)	31 (68.9)	
Māori	9 (19.6)	7 (15.6)	
Pacific	5 (10.9)	1 (2.2)	
Other	5 (10.9)	6 (13.3)	

Table 6. Baseline demographic characteristics

Values are expressed as mean \pm SD or as percentages. BMI = body mass index; RYGB = Rouxen-Y gastric bypass; SG = sleeve gastrectomy.

	$\mathbf{DVCP}(\mathbf{n}=46)$	SC(n-45)	Dyalua
DW (lag)	KIGD (II-40)	5G (II-45)	r value
BW (Kg)	112.7(12.0)	1155(122)	0.(2
	$113.7 (\pm 2.9)$	$113.3 (\pm 3.2)$	0.03
1-Year	81.8 (±2.6)	$8/./(\pm 2.1)$	0.12
5-Year	88.6 (±2.7)	101.4 (±2.4)	<0.001
%EWL			
l-Year	85.8 (±5.0)	65.9 (±3.2)	< 0.001
5-Year	66.8 (±4.2)	32.4 (±2.7)	<0.0001
TBF (kg)			
Baseline	51.3 (±1.9)	52.2 (±2.1)	0.72
1-Year	25.3 (±1.5)	30.2 (±1.4)	0.037
5-Year	33.4 (±1.5)	43.3 (±1.6)	< 0.0001
TBF%			
Baseline	45.0 (±1.1)	44.9 (±1.1)	0.93
1-Year	30.5 (±1.2)	34.4 (±1.3)	0.016
5-Year	37.6 (±1.1)	42.7 (±1.1)	0.002
FFM (kg)			
Baseline	62.5 (±1.9)	63.6 (±1.9)	0.68
1-Year	57.3 (±1.9)	58.2 (±1.8)	0.74
5-Year	55.6 (±1.9)	58.6 (±1.8)	0.26
FFM%			
Baseline	55.1 (±1.1)	55.3 (±1.1)	0.89
1-Year	70.5 (±1.3)	66.4 (±1.3)	0.017
5-Year	62.9 (±1.2)	57.9 (±1.1)	0.003
ASMM (kg)			
Baseline	26.0 (±0.9)	26.9 (±0.9)	0.51
1-Year	$22.3(\pm 0.9)$	$23.0(\pm 0.8)$	0.56
5-Year	$21.8 (\pm 0.9)$	$24.0(\pm 0.9)$	0.095
ASMM%		()	
Baseline	$22.8(\pm 0.5)$	$23.3(\pm 0.5)$	0.59
1-Year	$27.0(\pm 0.5)$	$26.2 (\pm 0.6)$	0.26
5-Year	$24.5 (\pm 0.6)$	$23.5(\pm 0.6)$	0.20
AFAT (kg)	2 110 (-010)	2010 (-010)	0.20
Baseline	5 57 (+0 22)	5 70 (+0 25)	0.64
1-Year	234 (+0.17)	$2.89 (\pm 0.16)$	0.05
5-Vear	3 31 (+0.19)	4 49 (+0.18)	<0.001
VFAT (kg)	$5.51(\pm 0.17)$	4.49 (±0.10)	\$0.001
Baseline	258(+017)	2 59 (+0 16)	0.98
1_Vear	$2.30 (\pm 0.17)$ 0.01 (±0.07)	$2.33 (\pm 0.10)$ 1 13 (± 0.08)	0.20
5 Vear	$1.20 (\pm 0.10)$	$1.13 (\pm 0.03)$ 1.08 (± 0.12)	<0.001
	1.27 (±0.17)	1.70 (±0.12)	<u>\0.001</u>
V FALATAL Deseline	471(+26)	A7 A (±2 7)	0.02
	(± 2.0)	$+/.+(\pm 2.7)$	0.70
1-itar 5 Veer	$42.7 (\pm 3.1)$	(± 2.7)	0.70
J-rear	37.2 (±2.3)	44.9 (±2.4 <i>)</i>	0.12

Table 7. Body weight and body composition measured by DXA at baseline, 1 year and 5 years.

Values are expressed as mean ± standard error of the mean. BW = body weight; %EWL = percentage excess weight loss; TBF = total body fat; FFM = fat-free mass; ASMM = appendicular skeletal muscle mass; AFAT = android fat; VFAT = visceral fat; VFAT:AFAT = (VFAT/AFAT)*100.



Figure 1. (A) Total changes in body weight (BW), total body fat (TBF), fat-free mass (FFM), and appendicular skeletal muscle mass (ASMM) after five years for RYGB (\square) and SG (\square) patients. (B) Patient body composition at five years as measured by DXA for RYGB (\square) and SG (\square). * P<0.05 for the difference between the groups.

4.2 Body Weight

Body weight (BW) decreased across the entire cohort from baseline to (T0) to 5-years (T2) and is shown in Table 8 and Figures 1 & 2A. Mixed model analysis detected a significant operation x time interaction (p<0.0001) and a statistically significant difference in BW at five years (mean difference: 12.79 kg, p=0.0009; Figures 1B & 2A). There was no significant difference in change in BW between the groups after one year (p=0.097). Between 1 and 5 years, weight regain was greater in the SG group than in the RYGB group (13.7 kg vs 6.8 kg, p<0.001) (p<0.001). At five years, the mean total weight loss in the RYGB group was 25.1 kg, compared to 14.1 kg in the SG group (p<0.0001).

Mixed model analysis also identified a significant operation x time interaction (p=0.0037) for %EWL, and differences in %EWL at one year (p=0.0005) and five years (p<0.0001) were seen (Figure 2B). In the RYGB group, %EWL was 85.8% at one year and 66.8% at five years. %EWL was significantly lower in the SG group at both time points, with %EWL of 65.9% at one year (p=0.001) and 32.4% at five years (p<0.0001).



Figure 2. (*A*) Body weight measured at baseline, one year and five years post-surgery for SG (Δ) and RYGB (\blacktriangle). (*B*) Percentage excess weight loss at one year and five years post-surgery for SG (Δ) and RYGB (\blacktriangle). %EWL was calculated as 100 * (Baseline Weight – Follow-up Weight)/(Baseline Weight – IBW).

Parameter	RYGB (n=46)	SG (n=45)	P value
$\Delta BW (kg)$			
1-year	-31.9 (±1.5)	-27.8 (±1.9)	0.097
5-year	-25.1 (±1.6)	-14.1 (±1.5)	< 0.0001
$\Delta 1$ -5 years	6.8 (±1.1)	13.7 (±1.6)	< 0.001
%EWL (%)			
1-year	85.8 (±5.0)	66.0 (±3.2)	0.001
5-year	66.8 (±4.2)	32.4 (±2.7)	< 0.0001
$\Delta 1$ -5 years	-18.9 (±3.3)	-33.3 (±3.6)	0.004

Table 8. Change in BW and %EWL over 5 years.

Values are expressed as mean \pm standard error of the mean. $\Delta BW =$ Change in body weight; %EWL calculated as: [(initial weight – current weight) / (initial weight – ideal weight)] × 100.

4.3 Fat Mass

As shown in Table 9, a greater reduction of FM occurred in the RYGB group over five years. Mixed model analysis identified a significant operation x time effect across the groups (p<0.001; Figure 3). Change in TBF at five years was -17.9 kg (7.4%) following RYGB compared to -8.8 kg (2.2%) in the SG group (Δ TBF; p<0.0001, Δ TBF%; p<0.0001). Fat loss occurred rapidly after surgery with reductions in TBF of 26 kg for RYGB and 21.9 kg for SG after one year (p=0.053). Following the short-term loss of FM after one year, the SG group regained significantly more FM between 1 and 5 years than the RYGB group (p= 0.003). All within-group changes were highly significant (p<0.0001).

Mixed model analysis also identified significant operation x time effects for AFAT (p<0.001), VFAT (p<0.001), and VFAT:AFAT (p=0.008) (Figure 4). Changes in central FM followed the same pattern as observed for TBF and are shown in Table 9. Changes in AFAT and VFAT at five years in the RYGB arm were -2.3 kg and -1.3 kg, respectively. For SG, smaller losses of 1.2 kg of AFAT and 0.6 kg of VFAT were seen (p<0.001 for both). Initial losses of both AFAT and VFAT at one year were not statistically different between the groups. However, between 1-year and 5-year follow-ups, the SG group regained a greater amount of AFAT (p=0.002) and VFAT (p<0.001).



Figure 3. (*A*) Total body fat measured by DXA at baseline, one year and five years post-surgery for SG (Δ) and RYGB (\blacktriangle). (*B*) Total body fat % at baseline, one year, and five years post-surgery for SG (Δ) and RYGB (\blacktriangle).







Figure 4. (A) Patient central fat distribution at five years as measured by DXA for RYGB (\blacksquare) and SG (\blacksquare). (B) Visceral fat (VFAT) measured by DXA at baseline, one year and five years post-surgery for SG (Δ) and RYGB (\blacktriangle). (C) Visceral fat as a percentage of android fat (AFAT) at baseline, one year, and five years post-surgery for SG (Δ) and RYGB (\blacktriangle). (D) Android fat (AFAT) at baseline, one year, and five years post-surgery for SG (Δ) and RYGB (\bigstar).

	RYGB (n=46)	SG (n=45)	P value
$\Delta TBF (kg)$			
1-year	-26.0 (±1.4)	-21.9 (±1.6)	0.053
5-year	-17.9 (±1.4)	-8.8 (±1.2)	< 0.0001
$\Delta 1$ -5 years	8.2 (±0.9)	13.1 (±1.3)	< 0.01
$\Delta TBF\%$ (%)			
1-year	-14.5 (±0.9)	-10.5 (±0.8)	0.001
5-year	-7.4 (±0.7)	$-2.2(\pm 0.5)$	< 0.0001
$\Delta 1$ -5 years	7.1 (±0.8)	8.3 (±0.7)	0.30
$\Delta AFAT$ (kg)			
1-year	-3.23 (±0.16)	-2.81 (±0.19)	0.10
5-year	-2.26 (±0.15)	-1.22 (±0.15)	< 0.0001
$\Delta 1$ -5 years	0.97 (±0.11)	1.60 (±0.15)	< 0.01
$\Delta VFAT$ (kg)			
1-year	-1.67 (±0.13)	-1.46 (±0.12)	0.23
5-year	-1.29 (±0.10)	-0.60 (±0.10)	< 0.0001
$\Delta 1$ -5 years	0.38 (±0.08)	0.85 (±0.10)	< 0.001
$\Delta VFAT:AFAT (\%)$			
1-year	-4.38 (±2.33)	-6.13 (±1.28)	0.60
5-year	-7.86 (±1.10)	-2.46 (±1.22)	< 0.01
$\Delta 1$ -5 years	-3.48 (±2.36)	3.67 (±1.49)	< 0.01

Table 9. Changes in total body fat and central fat mass over 5 years

Values are expressed as mean \pm standard error of the mean. P values derived from Student's ttest. TBF = total body fat, TBF% = body fat percentage, AFAT = android fat, VFAT = visceral fat, VFAT:AFAT = 100*(VFAT/AFAT).

4.4 Fat-Free Mass

Alongside reductions in TBF, both surgery groups had significant losses in FFM after five years, with reductions of 6.9 kg in RYGB (p<0.0001) and 5.0 kg in SG (p<0.0001), as shown in Table 10. Mixed model analysis confirmed significant operation x time interactions for FFM (p=0.0015), FFM% (p<0.001), ASMM (p=0.0012) and ASMM% (p=0.0027) (Figure 5).

Loss of FFM primarily occurred during the first year following surgery in both groups. However, a further gradual loss of 1.7 kg of FFM occurred beyond one year in the RYGB group, while the SG group had a minor increase in FFM of 0.4 kg. Despite greater losses of absolute FFM at five years, FFM% was higher in RYGB than SG (62.9% versus 57.9%; p=0.003). A greater increase in FFM% was seen at one year (p<0.001) and five years (p<0.0001) in RYGB than in SG (Table 10).

Trends in FFM were reflected in changes in ASMM. Both groups had similar absolute reductions in ASMM at one year (p=0.75) that equated to a greater increase in ASMM% in RYGB than SG

(p=0.003). Additionally, further loss of ASMM occurred between 1 and 5 years in the RYGB group (-0.5 kg) and not in the SG group (+0.9 kg). No significant differences were found between the groups for ASMM or ASMM% at five years (Table 7). ASMM trended towards significance (p=0.095), and analysis of changes over five years in ASMM and ASMM% showed significantly greater changes for RYGB (Table 10).

	RYGB (n=46)	SG (n=45)	P value	
Δ FFM (kg)				
1-year	-5.2 (±0.4)	-5.4 (±0.4)	0.73	
5-year	$-6.9(\pm 0.5)$	-5.0 (±0.5)	< 0.01	
$\Delta 1$ -5 years	-1.7 (±0.5)	$0.4~(\pm 0.4)$	< 0.001	
$\Delta FFM\%$ (%)				
1-year	15.4 (±0.9)	11.1 (±0.8)	< 0.001	
5-year	7.9 (±0.8)	2.6 (±0.5)	< 0.0001	
$\Delta 1$ -5 years	-7.5 (±0.9)	-8.6 (±0.8)	0.39	
$\Delta ASMM$ (kg)				
1-year	-3.8 (±0.2)	-3.9 (±0.3)	0.75	
5-year	-4.2 (±0.4	-2.9 (±0.3)	< 0.01	
$\Delta 1$ -5 years	-0.5 (±0.3)	0.9 (±0.2)	0.001	
Δ ASMM% (%)				
1-year	4.2 (±0.3)	2.9 (±0.3)	< 0.01	
5-year	1.7 (±0.4)	0.3 (±0.2)	< 0.01	
$\Delta 1$ -5 years	-2.5 (±0.3)	-2.6 (±0.3)	0.79	

Table 10. Changes in fat-free mass and appendicular skeletal muscle mass over 5 years

Values are expressed as mean \pm standard error of the mean. P values derived from Student's ttest. FFM = fat-free mass. ASMM = appendicular skeletal muscle mass.



Figure 5. Fat-free tissue measured by DXA at baseline, 1 year and 5 years for SG (Δ) and RYGB (\blacktriangle). (A) Fat-free mass (FFM). (B) Fat-free mass as a percentage of body weight. (C) Appendicular skeletal muscle mass (ASMM). (D) Appendicular skeletal muscle mass as a percentage of body weight.

4.5 Bioimpedance Equation Performance

A literature search was completed for pre-existing SF-BIA equations suitable for retrospective application to the study cohort. Three equations were selected for analysis (Table 11) in addition to factory equations from the single-frequency and multi-frequency devices utilised during data collection. Body composition at baseline was estimated for 88 participants with complete bioimpedance data. The surgery groups were combined for the bioimpedance analysis. Four equations calculated TBF, and one calculated FFM. TBF and FFM were subsequently derived for each equation using the assumption that TBF + FFM = BW.

Of the included equations, only one was derived from a population with obesity (Horie et al., 2008). The other equations of Deurenberg et al. (1991) and Roubenoff et al. (1997) have previously been applied to a population with obesity in a 2022 study (Coëffier et al., 2022). The Roubenoff equation was found to be the most accurate for predicting body composition in that study.

Author	Population	Reference Method	Equation
Deurenberg et al., 1991	Healthy Subjects	Densitometry	$FFM = -12.44 + (0.34 * h^2/R_{50}) + (0.1534 * h) + (0.273 * Wt) - (0.127 * age) + (4.229 * sex)$
Horie et al., 2008	Bariatric Patients	Densitometry	$TBF = 23.25 + (0.13 * age) + Wt + (0.09 * R_{50}) - (0.80 * h)$
Roubenoff et al., 1997	Elderly People	DXA	$FFM = 5.741 + (0.4551 * h^2/R_{50}) + (0.1405 * Wt) + (0.0573 * X_{50}) + (6.2467 * sex)$

Values are expressed as mean \pm standard error of the mean. P values derived from Student's ttest. FFM = fat-free mass. ASMM = appendicular skeletal muscle mass.

Comparative data between BIA and DXA-derived values for TBF and FFM are reported in Table 12 and Figure 6. Pearson correlation r was greater than 0.9 for all BIA equations for TBF. The lowest r was found in both factory equations (0.94) and the highest for the Deurenberg equation (0.98). The highest Lin concordance coefficient of 0.96 was also observed for the Deurenberg equation. The Deurenberg equation showed the smallest bias (-1.02 kg, -0.88%), and the Roubenoff equation had the largest bias (4.93 kg, 10.23%). Large biases were also seen for the BIA-101 and Quadscan 4000

devices. Only the Deurenberg equation surpassed an accuracy of 50% for a margin of \pm 5%. Large limits of agreement were observed for all equations, as shown in Bland-Altman plots (Figure 6) and Table 13. However, a strong proportional bias was evident for the Deurenberg equation with BIA overestimating TBF at TBF<50 kg with underestimation at higher TBF. Significant proportional biases were also seen for the Horie and BIA-101 equations.

Findings for FFM were similar to TBF. Pearson correlation r was equal to or greater than 0.95 in all equations. The Lin concordance coefficient ranged from 0.88 to 0.96. Accuracy was slightly higher for FFM as four equations exceeded 75% accuracy for a margin of $\pm 10\%$. The Deurenberg equation had the lowest bias (0.88 kg, 1.47%). However, large limits of agreement were found for all five equations (Figure 6 and Table 13). No proportional bias was seen for the Deurenberg and Horie equations, while the others all showed significant proportional bias.

Method	Fat mass (kg)	Fat-free mass (kg)	—
DXA	51.8 ± 1.4	62.6 ±1.3	
Deurenberg equation	50.7 ± 1.2	63.5 ±1.3	
Horie equation	47.9 ± 1.5	66.3 ±1.3	
Roubenoff equation	56.7 ± 1.4	57.6 ±1.2	
RJL Systems BIA-101	47.8 ± 1.2	66.4 ± 1.6	
Quadscan-4000	49.0 ± 1.5	65.2 ± 1.6	

Table 12. Fat mass and fat-free mass obtained by DXA and BIA equations.

Values are expressed as mean \pm standard error of the mean.

	Bias	95% CI (kg)	Bias (%)	Accuracy (%)		Pearson r	Lin p
	(kg)			$\pm 5\%$	$\pm 10\%$		
Fat mass							
Deurenberg et al.	-1.02	[-1.73; -0.30]	-0.88	54.6	87.5	0.98	0.96
Horie et al.	-3.83	[-4.65; -3.02]	-8.20	26.1	61.4	0.96	0.91
Roubenoff et al.	4.93	[4.31; 5.55]	10.23	17.1	56.8	0.97	0.89
RJL Systems BIA-101	-3.92	[-4.01; -2.93]	-6.76	33.0	59.1	0.94	0.90
Quadscan-4000	-2.72	[-3.73; -1.72]	-5.59	40.9	63.6	0.94	0.92
Fat-free mass							
Deurenberg et al.,	0.88	[0.15; 1.61]	1.47	63.6	92.1	0.96	0.96
Horie et al.	3.70	[2.92; 4.47]	6.21	35.2	77.3	0.96	0.92
Roubenoff et al	-5.06	[-5.66; -4.46]	-8.03	22.7	68.2	0.97	0.91
RJL Systems BIA-101	3.79	[2.79; 4.78]	5.78	43.2	76.1	0.95	0.88
Quadscan-4000	2.59	[1.61; 3.57]	3.78	45.5	81.8	0.95	0.92

Table 13. Comparison of fat mass and fat-free mass obtained by DXA and BIA equations.

Bias was calculated as: BIA calculated value– DXA value. 95% CI, 95% confidence interval. Accuracy is the percentage of participants with bias values within $\pm 5\%$ or $\pm 10\%$ of DXA value.



DXA (kg)

Average [(DXA + BIA)/2] in kg

В **Horie Equation** Fat Mass Difference [BIA - DXA] in kg Horie Equation (kg) 1,965 C -10 e Mean - 1.96S r=0.30 p=0.005 -20 DXA (kg) Average [(DXA + BIA)/2] in kg

Fat-Free Mass







C

BIA-SF Factory Equation (RJL Systems Model BIA-101)

D



46

BIA-MF Factory Equation (Bodystat Ltd, Quadscan 4000)



Figure 6. Scatter graphs and Bland-Altman plots for the comparison of fat mass and fat-free mass measured by DXA and BIA-derived values from equations. On the left-hand graphs, regression lines are shown with 95% confidence limits for the mean (blue dashed lines) and the line of identity (red dashed line). On the Bland-Altman plots, regression lines are shown in red, along with Pearson correlation coefficients (r).

Chapter 5. Discussion

Weight loss after bariatric surgery is known to be highly variable and dependent on factors such as pre-operative weight, the presence of comorbidities, and post-operative dietary habits (Livhits et al., 2011). Consideration of this variability is critical when assessing the currently available literature on bariatric surgery outcomes. The current evidence base is heavily informed by observational studies that are inherently susceptible to internal error due to their methodology. Furthermore, investigations into body composition outcomes beyond the initial twelve months post-surgery are scarce. To our knowledge, this is the first and only randomised control trial to use DXA to assess body composition outcomes five years after bariatric surgery. While some studies have investigated the long-term outcomes of bariatric surgery, these generally were observational in nature and lacked the use of gold-standard techniques for body composition analysis (Table 5). The follow-up duration, relatively large sample size, and use of DXA to assess body composition make this study novel.

5.1 Body Weight

Weight loss at one year measured as %EWL in RYGB (85.8%) and SG (65.9%) exceeded findings reported in studies by Otto et al. (2015) and Kavanagh et al. (2019). In a prospective, non-randomised study of 173 participants, Otto et al. (2015) found % an EWL of 62.9% in RYGB and 52.3% in SG. The observed difference in %EWL reported by Otto et al. may be a result of differences in baseline characteristics as %TWL was similar between studies. %TWL at one year was 28.2% (RYGB) and 23.3% (SG) within our cohort, and 31.7% (RYGB) and 30.4% (SG) in Otto et al. Therefore, discrepancies in %EWL may reflect differences in baseline BMI as group means were notably lower within our cohort (RYGB: 39.0±5.2 kg/m² and SG: 40±6.0 kg/m² compared to RYGB: 45.6±5.7 kg/m² and SG: 55.9±7.8 kg/m²). Another prospective, non-randomised study of 66 individuals by Kavanagh et al. (2019) observed %EWL of 53.4% in RYGB and 47.2% in SG. Baseline body weight and TWL at one year were similar to those of our cohort. However, baseline BMI was not published, and weight loss was only reported as TWL and %EWL. Additionally, no standard deviations or other measure of error were included. Therefore, it is unclear if within-group differences in %EWL.

%EWL at five years following RYGB (66.8%) was only slightly lower than in other long-duration studies. %EWL five years following RYGB was 76.2% in a prospective, randomised trial by Zhang et al. (2014) and 81.7% in a retrospective, non-randomized study by Ceriani et al. (2021). However,

the 32.4% %EWL in the SG group of our study is vastly inconsistent with the weight loss observed within SG cohorts in these two studies (Ceriani et al., 78.6%; Zhang et al., 62.35%). The disagreement between findings appears to develop throughout the study duration. In contrast with comparisons against shorter duration studies, %EWL of 65.9% was lower than observed by Zhang et al. (73.9%). This difference appears to have been confounded by notably greater weight regain between 1-year and 5-year measurements, with approximately half of the weight loss observed at one year being regained. This differs drastically from the current literature, with mean weight regain not typically exceeding 25% (King et al., 2020). The occurrence of excessive weight regain is further abnormal as its impact was isolated to the SG group. Excessive weight regain can be attributed to procedural failures but is more frequently a result of lifestyle factors such as inadequate physical activity and dysregulated eating (Noria et al., 2023).

Significant differences in weight loss were found between the two surgery groups at 1 and 5 years. Minor differences were identified at one year as weight loss was significant by %EWL and not by absolute weight despite similar baseline weights and BMIs. Differences in absolute weight loss and %EWL at five years were stark as weight regain between 1 and 5 years was significantly larger in the SG group. This suggests that weight loss following RYGB is better sustained than SG. Reported differences between weight loss outcomes at one year following RYGB and SG are often not statistically significant. No studies identified during the literature review process of this thesis reported differences in weight loss at one year that were not abolished by adjustment for differences in baseline characteristics (Table 4). In this context, the differences identified at one year appear abnormal. However, body composition changes are not isolated to the first-year post-bariatric surgery. Findings from studies with short follow-up durations are not necessarily reflective of the long-term outcomes.

The current literature extensively covers the immediate outcomes of bariatric surgery but fails to assess long-term changes. Pooled data by Uhe et al. (2021) and findings from long-term studies suggest that differences in weight loss and body composition may develop outside of the first 12 months (Yang et al., 2015; Zhang et al., 2014). Two randomised trials have reported on yearly body weight assessments following RYGB and SG. The study by Yang et al. (2015) monitored body weight in 64 individuals for three years and found differences in %TWL, TWL, and %EWL only after three years. Zhang et al. (2014) produced similar findings over five years in a sample of 64 individuals. Weight loss measured by %EWL was similar after one year but significantly greater

following RYGB at each yearly assessment between 2 and 5 years. Potential differences in weight loss at one year are likely marginal in comparison to the changes occurring beyond this point.

5.2 Fat-free mass

Few studies report relative changes in FFM following bariatric surgery. The pool of literature is further limited by non-standardized reporting methods. FFM and LBM are often mistakenly used interchangeably. Unlike FFM, LBM does not include bone mass (Müller et al., 2016). This thesis opted to report on FFM and %FFM as changes in bone mass following bariatric surgery correlate with changes in FFM and are unlikely to significantly contribute to changes in body weight (Brzozowska et al., 2013). A 2021 study retrospectively found no differences in change in FFM after five years between SG and RYGB patients (Ceriani et al., 2021). However, the study used BIA to determine FM and FFM. Despite appeals to its validity, bioimpedance techniques have not yet been shown to accurately measure body composition or its changes over time (Table 3).

Another 2018 study by Davidson et al. assessed body composition over five years using a 3compartment model consisting of total body density (BodPod) and TBW (D₂O dilution). An additional subset of the study cohort underwent whole-body MRI to measure skeletal muscle mass. During the first year, the RYGB group had larger reductions in FFM and greater increases in FFM%. After five years, the FFM% in RYGB was higher than in SG. The study was nonrandomised, and the RYGB group had a lower BMI and higher FFM% than the SG group at baseline. Additionally, the sample was unbalanced (RYGB; n=58, SG; n=11) and 5-year data was only available for 62% of recruited participants. Despite limitations in study methodology, trends in FFM observed by Davidson et al. are consistent with our observations. During the first year, both surgery groups have similar reductions in FFM alongside a larger increase in FFM% in RYGB. In both studies, FFM is maintained in SG between 1 and 5 years, while a mild decrease is observed in RYGB. Despite further loss in FFM, FFM% remained higher in RYGB than SG.

This trend emphasises the importance of including relative measures of body composition. Despite greater loss of FFM, increases in FFM% were significantly larger in RYGB than SG. FFM losses are known to strongly correlate with reductions in body weight. Therefore, differences in FFM between RYGB and SG are likely attributable to differences in weight loss and fat loss.

5.3 Fat Mass

Significant reductions in TBF were observed across the entire cohort during the first year. TBF and TBF% were lower in the RYGB arm at both one year and five years. Fat loss within the first year did not reach statistical significance (Table 9). However, reductions in TBF and TBF% were significantly greater at five years in the RYGB arm. Trends in FM were consistent with those observed in body weight and FFM as the SG group regained a greater amount of FM between one year and five years. Again, these findings are inconsistent with the pre-existing literature.

Studies by Kavanagh et al. (2019), Keidar et al. (2013), and Wells et al. (2015) all found no differences in the percentage change of FM between RYGB and SG after one year. One non-randomised trial reported a lower body fat percentage following RYGB, but this effect was abolished following adjustment for differences in baseline BMI (Otto et al., 2015). The previously mentioned studies by Ceriani et al. (2021) and Davidson et al. (2018) also found no differences in FM trends between RYGB and SG over five years.

5.4 Fat Mass Distribution

Changes in FM distribution are of particular importance due to the association between central obesity and metabolic disease (Kissebah et al., 1982; Lopes et al., 2016; Stefan, 2020). The effects of bariatric surgery on T2DM management in this study cohort have previously been published (Murphy et al., 2022). T2DM remission rates defined as an HbA_{1C} <6% or 42 mmol/mol were notably higher following RYGB than SG. The finding is not restricted to this cohort, as superior T2DM remission rates have been reported by other studies (Guerrero-Pérez et al., 2019; Kashyap et al., 2013). Differences in T2DM remission rates failed to reach significance in a systematic review by Sharples and Mahawar (2019), as resolution rates of 37.4% and 27.5% were found for RYGB and SG, respectively. Despite failing to reach significance, further investigation into the potential role of changes in central obesity is warranted. Correlations between abdominal adiposity and insulin sensitivity are well established but have not been investigated in the context of bariatric surgery (Goodpaster et al., 1997).

Android fat mass (AFAT) and visceral fat mass (VFAT) were lower in the RYGB group after five years. Analysis of the changes in central fat stores between assessment points reveals critical differences between one and five years. Reductions in both AFAT and VFAT after one year were similar between the groups. However, the SG regained significantly more AFAT and VFAT in five

years. Differences in weight regain appear as the primary contributor to differences after five years. This suggests that the potential beneficial effects of bariatric surgery on T2DM management may emerge over a longer time course.

5.5 Bioimpedance Equations

FM and FFM values measured by DXA were compared with values derived from four SF-BIA equations and one MF-BIA equation. Bias was small, ranging from -5.06 kg to 4.93 kg. Four equations showed bias that underestimated FM and overestimated FFM compared to DXA values. Only the Roubenoff equation overestimated FM and underestimated FFM. Investigations into bioimpedance methods consistently report the overestimation of FFM and underestimation of FM (Ballesteros-Pomar et al., 2022; Berstad et al., 2011; Faria et al., 2014; Verdich et al., 2011). In a study of women undergoing gastric banding, a strong correlation was found between values derived from SF-BIA and DXA. This correlation increased with weight loss, suggesting that the accuracy of SF-BIA is influenced negatively by body weight (Savastano et al., 2009). This is consistent with findings from Achamrah et al. (2018), where bias increased with BMI in participants with a BMI greater than 40 kg/m².

This highlights the need for specific equations developed for populations classified as overweight/obese. One of the equations included in the analysis was derived from a bariatric population (Horie et al., 2008). However, it had lower accuracy than the included factory equations. The performance of the Roubenoff equation was also inconsistent with previous findings. Following retrospective analysis, the Roubenoff equation was recommended for use in populations with obesity (Coëffier et al., 2022). The Roubenoff equation had a 5% accuracy of 53.1% for FFM and 34.8% for FM. However, in this study, the Roubenoff equation had the greatest bias and lowest accuracy for FM and FFM. Accuracy at 5% for the Roubenoff equation in this study was 22.7% for FFM and 17.1% for FM.

Despite the origins of a given equation, it must be validated in a population before considering its use. In this study cohort, the Deurenberg equation exhibited the lowest bias and highest accuracy for both FM and FFM. The correlation was high for all equations with Pearson coefficients above 0.94. However, large limits of agreement were found for all equations, which is evident in the Bland-Altman plots. This, alongside low accuracy, shows that SF-BIA and MF-BIA equations are not suitable for assessing body composition in individuals. SF-BIA and MF-BIA are appealing

alternatives for body composition assessment due to their non-invasive methodology and low costs. However, its validity in populations with obesity has not yet been proven.

Chapter 6. Conclusion

This study identified clear differences between body composition outcomes following RYGB and SG. These findings were mostly inconsistent with the results of other studies. RYGB produced favourable changes in body composition compared to SG. These findings challenge the consensus developed mostly from studies with follow-up durations of less than two years. Differences between the groups at five years were likely a product of minor short-term differences and more significant differences in long-term weight regain. To confirm the validity of these results, future randomised controlled trials with sufficiently long follow up periods are required.

References

- Abdeen, G., & Le Roux, C. W. (2015). Mechanism underlying the weight loss and complications of Roux-en-Y gastric bypass. review. Springer Science and Business Media LLC. 10.1007/s11695-015-1945-7
- Achamrah, N., Colange, G., Delay, J., Rimbert, A., Folope, V., Petit, A., Grigioni, S., Déchelotte, P., & Coëffier, M. (2018). Comparison of body composition assessment by DXA and BIA according to the body mass index: A retrospective study on 3655 measures. Public Library of Science. <u>https://doi.org/10.1371/journal.pone.0200465</u>
- Andreoli, A., Garaci, F., Cafarelli, F. P., & Guglielmi, G. (2016). Body composition in clinical practice. *European Journal of Radiology*, *85*(8), 1461-1468. 10.1016/j.ejrad.2016.02.005
- Bacchi, E., Cavedon, V., Zancanaro, C., Moghetti, P., & Milanese, C. (2017). Comparison between dual-energy X-ray absorptiometry and skinfold thickness in assessing body fat in overweigh/obese adult patients with type-2 diabetes. *Scientific Reports*, 7(1), 17424-8. 10.1038/s41598-017-17788-y
- Ballesteros-Pomar, M. D., González-Arnáiz, E., Pintor-De-La Maza, B., Barajas-Galindo, D., Ariadel-Cobo, D., González-Roza, L., & Cano-Rodríguez, I. (2022). *Bioelectrical impedance* analysis as an alternative to dual-energy x-ray absorptiometry in the assessment of fat mass and appendicular lean mass in patients with obesity. Elsevier BV. 10.1016/j.nut.2021.111442
- Baron, A. D., Brechtel, G., Wallace, P., & Edelman, S. V. (1988). Rates and tissue sites of noninsulin- and insulin-mediated glucose uptake in humans. *American Journal of Physiology-Endocrinology and Metabolism*, 255(6), 769. 10.1152/ajpendo.1988.255.6.e769
- Baumgartner, R. N., Heymsfield, S. B., & Roche, A. F. (1995). *Human body composition and the epidemiology of chronic disease*. Wiley. 10.1002/j.1550-8528.1995.tb00124.x
- Bazzocchi, A., Ponti, F., Albisinni, U., Battista, G., & Guglielmi, G. (2016). DXA: Technical aspects and application. *European Journal of Radiology*, *85*(8), 1481-1492. 10.1016/j.ejrad.2016.04.004
- Becroft, L., Ooi, G., Forsyth, A., King, S., & Tierney, A. (2019). Validity of multi-frequency bioelectric impedance methods to measure body composition in obese patients: A systematic review. *International Journal of Obesity*, 43(8), 1497-1507. 10.1038/s41366-018-0285-9
- Behnke, A. R., Feen, B. G., & Welham, W. C. (1942). The specific gravity of healthy men: Body weight ÷ volume as an index of obesity. *Journal of the American Medical Association*, 118(7), 495-498. 10.1001/jama.1942.02830070001001
- Benítez Brito, N., Suárez Llanos, J. P., Fuentes Ferrer, M., Oliva García, J. G., Delgado Brito, I., Pereyra-García Castro, F., Caracena Castellanos, N., Acevedo Rodríguez, C. X., & Palacio Abizanda, E. (2016). Relationship between mid-upper arm circumference and body mass index in inpatients. *PLoS ONE*, 11(8), e0160480. 10.1371/journal.pone.0160480
- Benito, P. J., Gómez-Candela, C., Cabañas, M. D., Szendrei, B., & Castro, E. A. (2019). Comparison between different methods for measuring body fat after a weight loss program. *Revista Brasileira De Medicina do Esporte, 25*(6)10.6084/m9.figshare.10295639.v1
- Bernhard, A. B., Santo, M. A., Scabim, V. M., Serafim, M. P., & de Cleva, R. (2016). Body composition evaluation in severe obesity: A critical review. Advances in Obesity, Weight Management & Control, 4(6)10.15406/aowmc.2016.04.00113

- Berstad, P., Randby, A., Seim Ekeland, G., Ulveland, H., Omland, T., & Almendingen, K. (2011). Body fat and fat-free mass measured by bioelectric impedance spectroscopy and dualenergy X-ray absorptiometry in obese and non-obese adults. Cambridge University Press (CUP). 10.1017/s000711451100417x
- Blüher, M., Aras, M., Aronne, L. J., Batterham, R. L., Giorgino, F., Ji, L., Pietiläinen, K. H., Schnell, O., Tonchevska, E., & Wilding, J. P. H. (2023). New insights into the treatment of obesity. *Diabetes, Obesity and Metabolism, 25*(8), 2058-2072. 10.1111/dom.15077
- Bohdjalian, A., Langer, F. B., Shakeri-Leidenmühler, S., Gfrerer, L., Ludvik, B., Zacherl, J., & Prager, G. (2010). Sleeve gastrectomy as sole and definitive bariatric procedure: 5-year results for weight loss and ghrelin. Springer Science and Business Media LLC. 10.1007/s11695-009-0066-6
- Bradley, D., Conte, C., Mittendorfer, B., Eagon, J. C., Varela, J. E., Fabbrini, E., Gastaldelli, A., Chambers, K. T., Su, X., Okunade, A., Patterson, B. W., & Klein, S. (2012). *Gastric bypass* and banding equally improve insulin sensitivity and β cell function. American Society for Clinical Investigation. 10.1172/jci64895
- Brožek, J., Grande, F., Anderson, J. T., & Keys, A. (1963). Densitometric analysis of body composition: Revision of some quantitative assumptions. *Annals of the New York Academy of Sciences*, *110*(1), 113-140. 10.1111/j.1749-6632.1963.tb17079.x
- Brzozowska, M., Sainsbury, A., Eisman, J. A., & Baldock, P. A. (2013). Bariatric surgery, bone loss, obesity and possible mechanisms. *Obesity Reviews*, *14*, 52-67. 10.1111/j.1467-789x.2012.01050.x
- Ceriani, V., Pinna, F., Galantino, A., Zakaria, A. S., Manfrini, R., Pontiroli, A. E., & Folli, F. (2021). Biliopancreatic diversion (BPD), long common limb revisional biliopancreatic diversion (BPD + LCL–R), roux-en-Y gastric bypass [RYGB] and sleeve gastrectomy (SG) mediate differential quantitative changes in body weight and qualitative modifications in body composition: A 5-year study. Springer Science and Business Media LLC. 10.1007/s00592-021-01777-9
- Chan, D. C., Watts, G. F., Barrett, P. H. R., & Burke, V. (2003). Waist circumference, waist-to-hip ratio and body mass index as predictors of adipose tissue compartments in men. *QJM*: *Monthly Journal of the Association of Physicians*, *96*(6), 441-447. 10.1093/qjmed/hcg069
- Cheng, J., Gao, J., Shuai, X., Wang, G., & Tao, K. (2016). The comprehensive summary of surgical versus non-surgical treatment for obesity: A systematic review and meta-analysis of randomized controlled trials. *Oncotarget*, 7(26), 39216-39230. 10.18632/oncotarget.9581
- Chooi, Y. C., Ding, C., & Magkos, F. (2019). *The epidemiology of obesity*. Elsevier BV. 10.1016/j.metabol.2018.09.005
- Chousleb, E., Rodriguez, J. A., & O'Leary, J. P. (2019). History of the Development of metabolic/bariatric surgery. *The ASMBS textbook of bariatric surgery* (pp. 37-46). Springer International Publishing. 10.1007/978-3-030-27021-6_3
- Clapp, B., Ponce, J., Demaria, E., Ghanem, O., Hutter, M., Kothari, S., Lamasters, T., Kurian, M., & English, W. (2022). American society for metabolic and bariatric surgery 2020 estimate of metabolic and bariatric procedures performed in the United States. Elsevier BV. 10.1016/j.soard.2022.06.284

- Coëffier, M., El Machkouri, M., L'Huillier, C., Folope, V., Lelandais, H., Grigioni, S., Déchelotte, P., & Achamrah, N. (2022). Accuracy of bioimpedance equations for measuring body composition in a cohort of 2134 patients with obesity. *Clinical Nutrition (Edinburgh, Scotland)*, 41(9), 2013-2024. 10.1016/j.clnu.2022.07.032
- Cotter, G., Metra, M., Milo-Cotter, O., Dittrich, H. C., & Gheorghiade, M. (2008). Fluid overload in acute heart failure - re-distribution and other mechanisms beyond fluid accumulation. *European Journal of Heart Failure*, 10(2), 165-169. 10.1016/j.ejheart.2008.01.007
- Cox-Reijven, P. L., van Kreel, B. K., & Soeters, P. B. (2002). Accuracy of bioelectrical impedance spectroscopy in measuring changes in body composition during severe weight loss. *JPEN*. *Journal of Parenteral and Enteral Nutrition*, 26(2), 120-127. 10.1177/0148607102026002120
- Crabtree, N. J., Leonard, M. B., & Zemel, B. S. (2007). Dual-energy X-ray absorptiometry. *Bone densitometry in growing patients* (pp. 41-57). Humana Press. 10.1007/978-1-59745-211-3
- Darsini, D., Hamidah, H., Notobroto, H. B., & Cahyono, E. A. (2020). Health risks associated with high waist circumference: A systematic review. *Journal of Public Health Research*, 9(2), jphr.2020.1811-1811. 10.4081/jphr.2020.1811
- Davidson, L. E., Yu, W., Goodpaster, B. H., Delany, J. P., Widen, E., Lemos, T., Strain, G. W., Pomp, A., Courcoulas, A. P., Lin, S., Janumala, I., Thornton, J. C., & Gallagher, D. (2018). Fat-Free mass and skeletal muscle mass five years after bariatric surgery. Wiley. 10.1002/oby.22190
- De Lorenzo, A., Andreoli, A., Matthie, J., & Withers, P. (1997). Predicting body cell mass with bioimpedance by using theoretical methods: A technological review. *Journal of Applied Physiology*, 82(5), 1542-1558. 10.1152/jappl.1997.82.5.1542
- Dee, A., Kearns, K., O'Neill, C., Sharp, L., Staines, A., O'Dwyer, V., Fitzgerald, S., & Perry, I. J. (2014). The direct and indirect costs of both overweight and obesity: A systematic review. *BMC Research Notes*, 7(1), 242. 10.1186/1756-0500-7-242
- Demerath, E. W., Guo, S. S., Chumlea, W. C., Towne, B., Roche, A. F., & Siervogel, R. M. (2002). Comparison of percent body fat estimates using air displacement plethysmography and hydrodensitometry in adults and children. *International Journal of Obesity*, 26(3), 389-397. 10.1038/sj.ijo.0801898
- Deurenberg, P., Kooy, K. v. d., Leenen, R., Weststrate, J. A., & Seidell, J. C. (1991). Sex and age specific prediction formulas for estimating body composition from bioelectrical impedance: A cross-validation study. *International Journal of Obesity*, 15(1), 17-25. <u>https://www.ncbi.nlm.nih.gov/pubmed/2010255</u>
- Durnin, J. V. G. A., & Rahaman, M. M. (1967). The assessment of the amount of fat in the human body from measurements of skinfold thickness. *British Journal of Nutrition*, 21(3), 681-689. 10.1079/BJN19670070
- Durnin, J. V. G. A., & Womersley, J. (1974). Body fat assessed from total body density and its estimation from skinfold thickness: Measurements on 481 men and women aged from 16 to 72 years. *British Journal of Nutrition*, *32*(1), 77-97. 10.1079/BJN19740060
- Earthman, C., Traughber, D., Dobratz, J., & Howell, W. (2007). Bioimpedance spectroscopy for clinical assessment of fluid distribution and body cell mass. SAGE Publications. 10.1177/0115426507022004389

- Ellis, K. J. (2000). Human body composition: In vivo methods. *Physiological Reviews*, 80(2), 649-680. 10.1152/physrev.2000.80.2.649
- Faria, S. L., Faria, O. P., Cardeal, M. D. A., & Ito, M. K. (2014). Validation study of multifrequency bioelectrical impedance with dual-energy X-ray absorptiometry among obese patients. Springer Science and Business Media LLC. 10.1007/s11695-014-1190-5
- Forbes, G. B., Gallup, J., & Hursh, J. B. (1991). Estimation of total body fat from potassium-40 content. *Nutrition Reviews*, 49(4), 109-111. 10.1111/j.1753-4887.1991.tb02998.x
- Gagner, M., Deitel, M., Erickson, A. L., & Crosby, R. D. (2013). Survey on laparoscopic sleeve gastrectomy (LSG) at the fourth international consensus summit on sleeve gastrectomy. Springer Science and Business Media LLC. 10.1007/s11695-013-1040-x
- Garrett, M., Poppe, K., Wooding, A., & Murphy, R. (2020). Private and public bariatric surgery trends in New Zealand 2004–2017: Demographics, cardiovascular comorbidity and procedure selection. *Obesity Surgery*, *30*(6), 2285-2293. 10.1007/s11695-020-04463-x
- GBD 2015 Obesity Collaborators, Afshin, A., Forouzanfar, M. H., Reitsma, M. B., Sur, P., Estep, K., Lee, A., Marczak, L., Mokdad, A. H., Moradi-Lakeh, M., Naghavi, M., Salama, J. S., Vos, T., Abate, K. H., Abbafati, C., Ahmed, M. B., Al-Aly, Z., Alkerwi, A., Al-Raddadi, R., . . . Murray, C. J. L. (2017). Health effects of overweight and obesity in 195 countries over 25 years. *The New England Journal of Medicine*, *377*(1), 13-27. 10.1056/NEJMoa1614362
- Gonçalves, S., Pecoits-Filho, R., Perreto, S., Barberato, S. H., Stinghen, A. E. M., Lima, E. G. A., Fuerbringer, R., Sauthier, S. M., & Riella, M. C. (2006). Associations between renal function, volume status and endotoxaemia in chronic kidney disease patients. *Nephrology, Dialysis, Transplantation, 21*(10), 2788-2794. 10.1093/ndt/gfl273
- Goodpaster, B. H., Thaete, F. L., Simoneau, J., & Kelley, D. E. (1997). Subcutaneous abdominal fat and thigh muscle composition predict insulin sensitivity independently of visceral fat. *Diabetes* (*New York, N.Y.*), 46(10), 1579-1585. 10.2337/diacare.46.10.1579
- Gray, D. S., Bray, G. A., Bauer, M., Kaplan, K., Gemayel, N., Wood, R., Greenway, F., & Kirk, S. (1990). Skinfold thickness measurements in obese subjects. *The American Journal of Clinical Nutrition*, 51(4), 571-577. 10.1093/ajcn/51.4.571
- Guerrero-Pérez, F., Casajoana, A., Gómez-Vaquero, C., Virgili, N., López-Urdiales, R., Hernández-Montoliu, L., Pujol-Gebelli, J., Osorio, J., Alves, C., Perez-Maraver, M., Pellitero, S., Vidal-Alabró, A., Fernández-Veledo, S., Vendrell, J., & Vilarrasa, N. (2019). *Changes in bone mineral density in patients with type 2 diabetes after different bariatric surgery procedures and the role of gastrointestinal hormones*. Springer Science and Business Media LLC. 10.1007/s11695-019-04127-5
- Guglielmi, G., Ponti, F., Agostini, M., Amadori, M., Battista, G., & Bazzocchi, A. (2016). The role of DXA in sarcopenia. *Aging Clinical and Experimental Research*, 28(6), 1047-1060. 10.1007/s40520-016-0589-3
- Guh, D. P., Zhang, W., Bansback, N., Amarsi, Z., Birmingham, C. L., & Anis, A. H. (2009). The incidence of co-morbidities related to obesity and overweight: A systematic review and meta-analysis. *BMC Public Health*, 9(1), 88. 10.1186/1471-2458-9-88
- Haas, V. K., Allen, J. R., Kohn, M. R., Clarke, S. D., Zhang, S., Briody, J. N., Gruca, M., Madden, S., Müller, M. J., & Gaskin, K. J. (2007). Total body protein in healthy adolescent girls:
 Validation of estimates derived from simpler measures with neutron activation analysis. *The American Journal of Clinical Nutrition*, 85(1), 66-72. 10.1093/ajcn/85.1.66

- Hattori, K., Numata, N., Ikoma, M., Danielson, R. R., & Matsuzaka, A. (1991). Sex differences in the distribution of subcutaneous and internal fat. *Human Biology*, 63(1), 53-63. <u>https://www.jstor.org/stable/41464142</u>
- Hewitt, M. J., Going, S. B., Williams, D. P., & Lohman, T. G. (1993). Hydration of the fat-free body mass in children and adults: Implications for body composition assessment. *American Journal of Physiology - Endocrinology and Metabolism, 265*(1), 88-95. 10.1152/ajpendo.1993.265.1.e88
- Heymsfield, S., Smith, R., Aulet, M., Bensen, B., Lichtman, S., Wang, J., & Pierson, R. (1990). Appendicular skeletal muscle mass: Measurement by dual-photon absorptiometry. *The American Journal of Clinical Nutrition, 52*(2), 214-218. 10.1093/ajcn/52.2.214
- Heymsfield, S. B., McManus, C., Smith, J., Stevens, V., & Nixon, D. W. (1982). Anthropometric measurement of muscle mass: Revised equations for calculating bone-free arm muscle area. *The American Journal of Clinical Nutrition*, 36(4), 680-690. 10.1093/ajcn/36.4.680
- Hill, J. O. (2006). Understanding and addressing the epidemic of obesity: An energy balance perspective. *Endocrine Reviews*, 27(7), 750-761. 10.1210/er.2006-0032
- Horie, L. M., Gonzalez Barbosa-Silva, M. C., Torrinhas, R. S., Túlio de Mello, M., Cecconello, I.,
 & Waitzberg, D. L. (2008). New body fat prediction equations for severely obese
 patients. *Clinical Nutrition (Edinburgh, Scotland)*, 27(3), 350-356. 10.1016/j.clnu.2008.03.011
- Houtkooper, L., Lohman, T., Going, S., & Howell, W. (1996). Why bioelectrical impedance analysis should be used for estimating adiposity. *The American Journal of Clinical Nutrition*, 64(3), 436S-448S. 10.1093/ajcn/64.3.436S
- Hu, H. H., Chen, J., & Shen, W. (2016). Segmentation and quantification of adipose tissue by magnetic resonance imaging. *Magma (New York, N.Y.), 29*(2), 259-276. 10.1007/s10334-015-0498-z
- Huang, R., Ding, X., Fu, H., & Cai, Q. (2019). Potential mechanisms of sleeve gastrectomy for reducing weight and improving metabolism in patients with obesity. Elsevier BV. 10.1016/j.soard.2019.06.022
- Institute of Medicine, & Committee on Military Nutrition Research. (1997). *Emerging technologies* for nutrition research : Potential for assessing military performance capability. National Academies Press.
- Jang, S., Park, J., Ryu, S., & Choi, S. (2020). Low muscle mass is associated with osteoporosis: A nationwide population-based study. Elsevier BV. 10.1016/j.maturitas.2020.01.003
- Karamanakos, S. N., Vagenas, K., Kalfarentzos, F., & Alexandrides, T. K. (2008). Weight loss, appetite suppression, and changes in fasting and postprandial ghrelin and peptide-YY levels after roux-en-Y gastric bypass and sleeve gastrectomy. Ovid Technologies (Wolters Kluwer Health). 10.1097/sla.0b013e318156f012
- Kashyap, S. R., Bhatt, D. L., Wolski, K., Watanabe, R. M., Abdul-Ghani, M., Abood, B., Pothier, C. E., Brethauer, S., Nissen, S., Gupta, M., Kirwan, J. P., & Schauer, P. R. (2013). *Metabolic effects of bariatric surgery in patients with moderate obesity and type 2 diabetes*. American Diabetes Association. 10.2337/dc12-1596
- Katch, F., Michael, E. D., & Horvath, S. M. (1967). Estimation of body volume by underwater weighing: Description of a simple method. *Journal of Applied Physiology (1948), 23*(5), 811-813. 10.1152/jappl.1967.23.5.811

- Kavanagh, R., Smith, J., Avgenackis, E., Jones, D., & Nau, P. (2019). A comparison of the effects of roux-en-Y gastric bypass and sleeve gastrectomy on body mass composition as measured by air displacement plethysmography. Springer Science and Business Media LLC. 10.1007/s11695-019-04178-8
- Kehayias, J., & Valtueña, S. (1999). Neutron activation analysis determination of body composition. *Current Opinion in Clinical Nutrition and Metabolic Care*, 2(6), 453-463. 10.1097/00075197-199911000-00004
- Keidar, A., Hershkop, K. J., Marko, L., Schweiger, C., Hecht, L., Bartov, N., Kedar, A., & Weiss, R. (2013). Roux-en-Y gastric bypass vs sleeve gastrectomy for obese patients with type 2 diabetes: A randomised trial. Springer Science and Business Media LLC. 10.1007/s00125-013-2965-2
- Kenngott, H. G., Nickel, F., Wise, P. A., Wagner, F., Billeter, A. T., Nattenmüller, J., Nabers, D., Maier-Hein, K., Kauczor, H., Fischer, L., & Müller-Stich, B. P. (2019). Weight loss and changes in adipose tissue and skeletal muscle volume after laparoscopic sleeve gastrectomy and roux-en-Y gastric bypass: A prospective study with 12-month follow-up. Springer Science and Business Media LLC. 10.1007/s11695-019-04087-w
- King, W. C., Hinerman, A. S., & Courcoulas, A. P. (2020). Weight regain after bariatric surgery: A systematic literature review and comparison across studies using a large reference sample. *Surgery for Obesity and Related Diseases*, 16(8), 1133-1144. 10.1016/j.soard.2020.03.034
- Kissebah, A. H., Vydelingum, N., Murray, R., Evans, D. J., Hartz, A. J., alkhoff, R. K., & Adams, P. W. (1982). Relation of body fat distribution to metabolic complications of obesity. *The Journal of Clinical Endocrinology and Metabolism*, 54(2), 254-260. 10.1210/jcem-54-2-254
- Korner, J., Punyanitya, M., Taveras, C., Mcmahon, D. J., Kim, H. J., Inabnet, W., Bessler, M., Gallagher, D., & Smith-Gordon, ©. (2008). Sex differences in visceral adipose tissue postbariatric surgery compared to matched non-surgical controls. *International Journal of Body Composition Research*, 6(3), 93-99. <u>https://www.ncbi.nlm.nih.gov/pubmed/20582247</u>
- Kushner, R., Schoeller, D., Fjeld, C., & Danford, L. (1992). Is the impedance index (ht2/R) significant in predicting total body water? *The American Journal of Clinical Nutrition*, 56(5), 835-839. 10.1093/ajcn/56.5.835
- Kushner, R. (1992). Bioeletrical impedance analysis: A review of principles and applications. *Journal of the American College of Nutrition*, 11(2), 199-209.
- Kyle, U. G., Bosaeus, I., De Lorenzo, A. D., Deurenberg, P., Elia, M., Gómez, J. M., Heitmann, B. L., Kent-Smith, L., Melchior, J., Pirlich, M., Scharfetter, H., Schols, A. M. W. J., & Pichard, C. (2004a). Bioelectrical impedance analysis—part I: Review of principles and methods. *Clinical Nutrition (Edinburgh, Scotland), 23*(5), 1226-1243. 10.1016/j.clnu.2004.06.004
- Kyle, U. G., Bosaeus, I., De Lorenzo, A. D., Deurenberg, P., Elia, M., Manuel Gómez, J., Lilienthal Heitmann, B., Kent-Smith, L., Melchior, J., Pirlich, M., Scharfetter, H., M.W.J Schols, A., & Pichard, C. (2004b). Bioelectrical impedance analysis—part II: Utilization in clinical practice. *Clinical Nutrition (Edinburgh, Scotland), 23*(6), 1430-1453. 10.1016/j.clnu.2004.09.012
- Laskey, M. A. (1996). *Dual-energy X-ray absorptiometry and body composition*. Elsevier Inc. 10.1016/0899-9007(95)00017-8

- Levitt, D. G., Beckman, L. M., Mager, J. R., Valentine, B., Sibley, S. D., Beckman, T. R., Kellogg, T. A., Ikramuddin, S., & Earthman, C. P. (2010). Comparison of DXA and water measurements of body fat following gastric bypass surgery and a physiological model of body water, fat, and muscle composition. *Journal of Applied Physiology*, 109(3), 786-795. 10.1152/japplphysiol.00278.2010
- Lin, L. I. (1989). A concordance correlation coefficient to evaluate reproducibility. *Biometrics*, 45(1), 255-268. 10.2307/2532051
- Livhits, M., Mercado, C., Yermilov, I., Parikh, J. A., Dutson, E., Mehran, A., Ko, C. Y., & Gibbons, M. M. (2011). *Preoperative predictors of weight loss following bariatric surgery: Systematic review*. Springer Science and Business Media LLC. 10.1007/s11695-011-0472-4
- Lohman, T. G., Roche, A. F., & Martorell, R. (1988). *Anthropometric standardization reference manual*. Human kinetics books.
- Lopes, H. F., Corrêa-Giannella, M. L., Consolim-Colombo, F. M., & Egan, B. M. (2016). Visceral adiposity syndrome. *Diabetology & amp; Metabolic Syndrome*, 8(1), 40. 10.1186/s13098-016-0156-2
- López-Taylor, J. R., González-Mendoza, R. G., Gaytán-González, A., Jiménez-Alvarado, J. A., Villegas-Balcázar, M., Jáuregui-Ulloa, E. E., & Torres-Naranjo, F. (2018). Accuracy of anthropometric equations for estimating body fat in professional male soccer players compared with DXA. *Journal of Sports Medicine, 2018*, 6843792-7. 10.1155/2018/6843792
- Machann, J., Thamer, C., Schnoedt, B., Haap, M., Haring, H., Claussen, C. D., Stumvoll, M., Fritsche, A., & Schick, F. (2005). Standardized assessment of whole body adipose tissue topography by MRI. *Journal of Magnetic Resonance Imaging*, 21(4), 455-462. 10.1002/jmri.20292
- Mager, J. R., Sibley, S. D., Beckman, T. R., Kellogg, T. A., & Earthman, C. P. (2008). Multifrequency bioelectrical impedance analysis and bioimpedance spectroscopy for monitoring fluid and body cell mass changes after gastric bypass surgery. *Clinical Nutrition* (*Edinburgh, Scotland*), 27(6), 832-841. 10.1016/j.clnu.2008.06.007
- Magkos, F., Fraterrigo, G., Yoshino, J., Luecking, C., Kirbach, K., Kelly, S. C., De las fuentes, L., He, S., Okunade, A. L., Patterson, B. W., & Klein, S. (2016). *Effects of moderate and* subsequent progressive weight loss on metabolic function and adipose tissue biology in humans with obesity. Elsevier BV. 10.1016/j.cmet.2016.02.005
- Marceau, P., Hould, F. -., Simard, S., Lebel, S., Bourque, R. -., Potvin, M., & Biron, S. (1998). *Biliopancreatic diversion with duodenal switch*
- Martin Bland, J., & Altman, D. (1986). Statistical methods for assessing agreement between two methods of clinical measurement. *The Lancet (British Edition), 327*(8476), 307-310. 10.1016/S0140-6736(86)90837-8
- Martinoli, R., Mohamed, E. I., Maiolo, C., Cianci, R., Denoth, F., Salvadori, S., & Iacopino, L. (2003). Total body water estimation using bioelectrical impedance: A meta-analysis of the data available in the literature. *Acta Diabetologica, 40 Suppl 1*(1), S203-s206. 10.1007/s00592-003-0066-2
- Matthie, J., Zarowitz, B., De Lorenzo, A., Andreoli, A., Katzarski, K., Pan, G., & Withers, P. (1998). Analytic assessment of the various bioimpedance methods used to estimate body water. *Journal of Applied Physiology*, 84(5), 1801-1816. 10.1152/jappl.1998.84.5.1801

- Matthie, J. R. (2008). *Bioimpedance measurements of human body composition: Critical analysis and outlook*. Informa Healthcare. 10.1586/17434440.5.2.239
- McCrory, M. A., Gomez, T. D., Bernauer, E. M., & Mole, P. A. (1995). Evaluation of a new air displacement plethysmograph for measuring human body composition. *Medicine and Science in Sports and Exercise*, 27(12), 1686-1691. 10.1249/00005768-199512000-00016
- Meredith-Jones, K., Haszard, J., Stanger, N., & Taylor, R. (2018). Precision of DXA-Derived visceral fat measurements in a large sample of adults of varying body size. *Obesity (Silver Spring, Md.), 26*(3), 505-512. 10.1002/oby.22108
- Mitsiopoulos, N., Baumgartner, R. N., Heymsfield, S. B., Lyons, W., Gallagher, D., & Ross, R. (1998). Cadaver validation of skeletal muscle measurement by magnetic resonance imaging and computerized tomography. *Journal of Applied Physiology*, 85(1), 115-122. 10.1152/jappl.1998.85.1.115
- Modi, P., Nasrin, S., Hawes, M., Glavis-Bloom, J., Alam, N. H., Hossain, M. I., & Levine, A. C. (2015). Midupper arm circumference outperforms weight-based measures of nutritional status in children with diarrhea. *The Journal of Nutrition*, 145(7), 1582-1587. 10.3945/jn.114.209718
- Moissl, U. M., Wabel, P., Chamney, P. W., Bosaeus, I., Levin, N. W., Bosy-Westphal, A., Korth, O., Müller, M. J., Ellegård, L., Malmros, V., Kaitwatcharachai, C., Kuhlmann, M. K., Zhu, F., & Fuller, N. J. (2006). Body fluid volume determination via body composition spectroscopy in health and disease. *Physiological Measurement*, 27(9), 921-933. 10.1088/0967-3334/27/9/012
- Moizé, V., Andreu, A., Rodríguez, L., Flores, L., Ibarzabal, A., Lacy, A., Jiménez, A., & Vidal, J. (2013). Protein intake and lean tissue mass retention following bariatric surgery. Elsevier BV. 10.1016/j.clnu.2012.11.007
- Montero, P. N., Stefanidis, D., Norton, H. J., Gersin, K., & Kuwada, T. (2011). *Reported excess* weight loss after bariatric surgery could vary significantly depending on calculation method: A plea for standardization. Elsevier BV. 10.1016/j.soard.2010.09.025
- Müller, M. J., Braun, W., Pourhassan, M., Geisler, C., & Bosy-Westphal, A. (2016). Application of standards and models in body composition analysis. *Proceedings of the Nutrition Society*, 75(2), 181-187. 10.1017/S0029665115004206
- Murphy, R., Plank, L. D., Clarke, M. G., Evennett, N. J., Tan, J., Kim, D. D. W., Cutfield, R., & Booth, M. W. C. (2022). Effect of banded roux-en-Y gastric bypass versus sleeve gastrectomy on diabetes remission at 5 years among patients with obesity and type 2 diabetes: A blinded randomized clinical trial. *Diabetes Care*, 45(7), 1503-1511. 10.2337/dc21-2498
- Noria, S. F., Shelby, R. D., Atkins, K. D., Nguyen, N. T., & Gadde, K. M. (2023). Weight regain after bariatric surgery: Scope of the problem, causes, prevention, and treatment. *Current Diabetes Reports*, 23(3), 31-42. 10.1007/s11892-023-01498-z
- Nuijten, M. A. H., Eijsvogels, T. M. H., Monpellier, V. M., Janssen, I. M. C., Hazebroek, E. J., & Hopman, M. T. E. (2022). The magnitude and progress of lean body mass, fat-free mass, and skeletal muscle mass loss following bariatric surgery: A systematic review and metaanalysis. *Obesity Reviews*, 23(1), e13370-n/a. 10.1111/obr.13370
- O'Brien, C., Young, A. J., & Sawka, M. N. (2002). Bioelectrical impedance to estimate changes in hydration status. *International Journal of Sports Medicine*, 23(5), 361-366. 10.1055/s-2002-33145

- Otto, M., Elrefai, M., Krammer, J., Weiß, C., Kienle, P., & Hasenberg, T. (2015). Sleeve gastrectomy and roux-en-Y gastric bypass lead to comparable changes in body composition after adjustment for initial body mass index. Springer Science and Business Media LLC. 10.1007/s11695-015-1792-6
- Pace, N., & Rathbun, E. (1945). The body water and chemically combined nitrogen content in relation to fat content. *J Biol Chem*, 158(3), 685-691.
- Park, B. S., & Yoon, J. S. (2013). Relative skeletal muscle mass is associated with development of metabolic syndrome. *Diabetes & Metabolism Journal*, 37(6), 458-464. 10.4093/dmj.2013.37.6.458
- Pierson Jr, R. N., Wang, J., Heymsfield, S. B., Russell-Aulet, M., Mazariegos, M., Tierney, M., Smith, R., Thornton, J. C., Kehayias, J., & Weber, D. A. (1991). Measuring body fat: Calibrating the rulers. intermethod comparisons in 389 normal caucasian subjects. *American Journal of Physiology - Endocrinology and Metabolism*, 261(1), 103-108. 10.1152/ajpendo.1991.261.1.E103
- Pietrobelli, A., Formica, C., Wang, Z., & Heymsfield, S. B. (1996). Dual-energy X-ray absorptiometry body composition model: Review of physical concepts. *American Journal of Physiology - Endocrinology and Metabolism, 271*(6), 941-951. 10.1152/ajpendo.1996.271.6.E941
- Plank, L. (2005). Dual-energy X-ray absorptiometry and body composition. *Current Opinion in Clinical Nutrition and Metabolic Care, 8*(3), 305-309. 10.1097/01.mco.0000165010.31826.3d
- Rinaldi, S., Gilliland, J., O'Connor, C., Chesworth, B., & Madill, J. (2019). Is phase angle an appropriate indicator of malnutrition in different disease states? A systematic review. *Clinical Nutrition ESPEN*, 29, 1-14. 10.1016/j.clnesp.2018.10.010
- Rothschild, P. A., Domesek, J. M., Eastham, M. E., & Kaufman, L. (1991). MR imaging of excessively obese patients: The use of an open permanent magnet. *Magnetic Resonance Imaging*, 9(2), 151-154. 10.1016/0730-725X(91)90003-5
- Roubenoff, R., Kehayias, J., Dawson-Hughes, B., & Heymsfield, S. (1993). Use of dual-energy x-ray absorptiometry in body-composition studies: Not yet a "gold standard". *The American Journal of Clinical Nutrition*, 58(5), 589-591. 10.1093/ajcn/58.5.589
- Roubenoff, R., Baumgartner, R. N., Harris, T. B., Dallal, G. E., Hannan, M. T., Economos, C. D., Stauber, P. M., Wilson, P. W. F., & Kiel, D. P. (1997). Application of bioelectrical impedance analysis to elderly populations. *The Journals of Gerontology. Series A, Biological Sciences and Medical Sciences*, 52A(3), M129-M136. 10.1093/gerona/52A.3.M129
- Ruban, A., Stoenchev, K., Ashrafian, H., & Teare, J. (2019). Current treatments for obesity. *Clinical Medicine (London, England), 19*(3), 205-212. 10.7861/clinmedicine.19-3-205
- Rubino, F., Schauer, P. R., Kaplan, L. M., & Cummings, D. E. (2010). *Metabolic surgery to treat type 2 diabetes: Clinical outcomes and mechanisms of action*. Annual Reviews. 10.1146/annurev.med.051308.105148
- Sahakyan, K. R., Somers, V. K., Rodriguez-Escudero, J. P., Hodge, D. O., Carter, R. E., Sochor, O., Coutinho, T., Jensen, M. D., Roger, V. L., Singh, P., & Lopez-Jimenez, F. (2015). Normalweight central obesity: Implications for total and cardiovascular mortality. *Annals of Internal Medicine*, 163(11), 827-835. 10.7326/M14-2525
- Sakai, S., Tanimoto, K., Imbe, A., Inaba, Y., Shishikura, K., Tanimoto, Y., Ushiroyama, T., Terasaki, J., & Hanafusa, T. (2016). *Decreased β-cell function is associated with reduced*

skeletal muscle mass in Japanese subjects without diabetes. Public Library of Science (PLoS). 10.1371/journal.pone.0162603

- Santanasto, A. J., Glynn, N. W., Newman, M. A., Taylor, C. A., Brooks, M. M., Goodpaster, B. H., & Newman, A. B. (2011). Impact of weight loss on physical function with changes in strength, muscle mass, and muscle fat infiltration in overweight to moderately obese older adults: A randomized clinical trial. Hindawi Limited. 10.1155/2011/516576
- Savastano, S., Belfiore, A., Di Somma, C., Mauriello, C., Rossi, A., Pizza, G., De Rosa, A., Prestieri, G., Angrisani, L., & Colao, A. (2009). Validity of bioelectrical impedance analysis to estimate body composition changes after bariatric surgery in premenopausal morbidly women. Springer Science and Business Media LLC. 10.1007/s11695-009-0006-5
- Sharples, A. J., & Mahawar, K. (2019). Systematic review and meta-analysis of randomised controlled trials comparing long-term outcomes of roux-en-Y gastric bypass and sleeve gastrectomy. Springer Science and Business Media LLC. 10.1007/s11695-019-04235-2
- Siri, W. E. (1956). Body composition from fluid spaces and density: Analysis of methods. University of California Research Laboratory(University of California Research Laboratory).
- Stefan, N. (2020). Causes, consequences, and treatment of metabolically unhealthy fat distribution. *The Lancet. Diabetes & Endocrinology*, 8(7), 616-627. 10.1016/S2213-8587(20)30110-8
- Strain, G. W., Gagner, M., Pomp, A., Dakin, G., Inabnet, W. B., Hsieh, J., Heacock, L., & Christos, P. (2009). Comparison of weight loss and body composition changes with four surgical procedures. Elsevier BV. 10.1016/j.soard.2009.04.001
- Tataranni, P., & Ravussin, E. (1995). Use of dual-energy X-ray absorptiometry in obese individuals. *The American Journal of Clinical Nutrition*, 62(4), 730-734. 10.1093/ajcn/62.4.730
- Taylor, J. L., Holland, D. J., Coombes, J. S., & Keating, S. E. (2021). Accuracy of dual-energy xray absorptiometry for assessing longitudinal change in visceral adipose tissue in patients with coronary artery disease. *International Journal of Obesity*, 45(8), 1740-1750. 10.1038/s41366-021-00840-3
- Thomas, E. L., Fitzpatrick, J. A., Malik, S. J., Taylor-Robinson, S. D., & Bell, J. D. (2013). Whole body fat: Content and distribution. *Progress in Nuclear Magnetic Resonance Spectroscopy*, 73, 56-80. 10.1016/j.pnmrs.2013.04.001
- Todorovic, V., Russell, C., Stratton, R., Ward, J., & Elia, M. (2011). *The 'MUST' explanatory booklet: A guide to the 'malnutrition universal screening tool' ('MUST') for adults*. British Association for Parenteral and Enteral Nutrition, BAPEN.
- Toombs, R. J., Ducher, G., Shepherd, J. A., & Souza, M. J. (2012). The impact of recent technological advances on the trueness and precision of DXA to assess body composition. *Obesity (Silver Spring, Md.), 20*(1), 30-39. 10.1038/oby.2011.211
- Toro-Ramos, T., Goodpaster, B. H., Janumala, I., Lin, S., Strain, G. W., Thornton, J. C., Kang, P., Courcoulas, A. P., Pomp, A., & Gallagher, D. (2015). Continued loss in visceral and intermuscular adipose tissue in weight-stable women following bariatric surgery. *Obesity (Silver Spring, Md.), 23*(1), 62-69. 10.1002/oby.20932
- Toso, S., Piccoli, A., Gusella, M., Menon, D., Bononi, A., Crepaldi, G., & Ferrazzi, E. (2000). Altered tissue electric properties in lung cancer patients as detected by bioelectric impedance

vector analysis. *Nutrition (Burbank, Los Angeles County, Calif.), 16*(2), 120-124. 10.1016/S0899-9007(99)00230-0

- Totosy de Zepetnek, J. O., Lee, J. J., Boateng, T., Plastina, S. E., Cleary, S., Huang, L., Kucab, M., Paterakis, S., Brett, N. R., & Bellissimo, N. (2021). Test–retest reliability and validity of body composition methods in adults. *Clinical Physiology and Functional Imaging*, 41(5), 417-425. 10.1111/cpf.12716
- Truesdale, K. P., Roberts, A., Cai, J., Berge, J. M., & Stevens, J. (2016). Comparison of eight equations that predict percent body fat using skinfolds in American youth. *Childhood Obesity*, 12(4), 314-323. 10.1089/chi.2015.0020
- Udelsman, B. V., Jin, G., Chang, D. C., Hutter, M. M., & Witkowski, E. R. (2019). Surgeon factors are strongly correlated with who receives a sleeve gastrectomy versus a roux-en-Y gastric bypass. Elsevier BV. 10.1016/j.soard.2019.03.040
- Uhe, I., Douissard, J., Podetta, M., Chevallay, M., Toso, C., Jung, M. K., & Meyer, J. (2022). Rouxen-Y gastric bypass, sleeve gastrectomy, or one-anastomosis gastric bypass? A systematic review and meta-analysis of randomized-controlled trials. Wiley. 10.1002/oby.23338
- Vahdat, K., Azizi, F., Zandi, K., Assadi, M., & Nabipour, I. (2012). *Chronic inflammation is correlated with percentage of body fat independent of the burden of infection*. Springer Science and Business Media LLC. 10.1007/s10753-012-9445-6
- Van Rijswijk, A., Van Olst, N., Schats, W., Van Der Peet, D. L., & Van De Laar, A. W. (2021). What is weight loss after bariatric surgery expressed in percentage total weight loss (%TWL)? A systematic review. Springer Science and Business Media LLC. 10.1007/s11695-021-05394-x
- Van Tonder, E., Mace, L., Steenkamp, L., Tydeman-Edwards, R., Gerber, K., & Friskin, D. (2019). Mid-upper arm circumference (MUAC) as a feasible tool in detecting adult malnutrition. *The South African Journal of Clinical Nutrition*, 32(4), 93-98. 10.1080/16070658.2018.1484622
- Verdich, C., Barbe, P., Petersen, M., Grau, K., Ward, L., Macdonald, I., Sørensen, T. I. A., & Oppert, J. (2011). Changes in body composition during weight loss in obese subjects in the NUGENOB study: Comparison of bioelectrical impedance vs. dual-energy X-ray absorptiometry. Elsevier BV. 10.1016/j.diabet.2010.10.007
- Waki, M. (. U., NY), Kral, J. G., Mazariegos, M., Wang, J., Pierson, R. N. J., & Heymsfield, S. B. (1991). Relative expansion of extracellular fluid in obese vs. nonobese women. *American Journal of Physiology - Endocrinology and Metabolism, 261*(2), 199-203. 10.1152/ajpendo.1991.261.2.e199
- Wang, J., Dilmanian, F. A., Thornton, J., Russell, M., Burastero, S., Mazariegos, M., Heymsfield, S. B., & Pierson, R. N. (1993). In vivo neutron activation analysis for body fat: Comparisons by seven methods. *Basic life sciences* (pp. 31-34). Springer US. 10.1007/978-1-4899-1268-8_5
- Wang, Y., Luo, D., Liu, J., Song, Y., Jiang, B., & Jiang, H. (2023). Low skeletal muscle mass index and all-cause mortality risk in adults: A systematic review and meta-analysis of prospective cohort studies. Public Library of Science (PLoS). 10.1371/journal.pone.0286745
- Wang, Z., Deurenberg, P., & Heymsfield, S. B. (2000). Cellular-level body composition model: A new approach to studying fat-free mass hydration. *Annals of the New York Academy of Sciences, 904*(1), 306-311. 10.1111/j.1749-6632.2000.tb06472.x
- Wells, J., Miller, M., Perry, B., Ewing, J. A., Hale, A. L., & Scott, J. D. (2015). Preservation of Fatfree Mass after Bariatric Surgery: A Comparison of Malabsorptive and Restrictive Procedures. *The American surgeon*, 81(8), 812–815.
- Widen, E. M., Strain, G., King, W. C., Yu, W., Lin, S., Goodpaster, B., Thornton, J., Courcoulas, A., Pomp, A., & Gallagher, D. (2014). Validity of bioelectrical impedance analysis for measuring changes in body water and percent fat after bariatric surgery. *Obesity Surgery*, 24(6), 847-854. 10.1007/s11695-014-1182-5
- World Health Organisation. (2000). Obesity: Preventing and managing the global epidemic. report of a WHO consultation. *World Health Organization Technical Report Series*, 894, i-253. http://europepmc.org/abstract/MED/11234459
- World Health Organisation. (2024). WHO factsheet: Obesity and overweight. (). https://www.who.int/news-room/fact-sheets/detail/obesity-and-overweight
- World Health Organization. (2011). Waist circumference and waist-hip ratio : Report of a WHO expert consultation, Geneva, 8-11 December 2008
- Yang, J., Wang, C., Cao, G., Yang, W., Yu, S., Zhai, H., & Pan, Y. (2015). Long-term effects of laparoscopic sleeve gastrectomy versus roux-en-Y gastric bypass for the treatment of Chinese type 2 diabetes mellitus patients with body mass index 28-35 kg/m2. Springer Science and Business Media LLC. 10.1186/s12893-015-0074-5
- Zhang, Y., Zhao, H., Cao, Z., Sun, X., Zhang, C., Cai, W., Liu, R., Hu, S., & Qin, M. (2014). A randomized clinical trial of laparoscopic roux-en-Y gastric bypass and sleeve gastrectomy for the treatment of morbid obesity in China: A 5-year outcome. Springer Science and Business Media LLC. 10.1007/s11695-014-1258-2